

ST. MARY'S UNIVERSITY SCHOOL OF GRADUATE STUDIES DEPARTMENT OF PROJECT MANAGEMENT

ASSESSMENT OF PROJECT MANAGEMENT PRACTICES IN HIGH-RISE BUILDING: THE CASE OF CBE HEADQUARTER BUILDING PROJECT

BY MAHIDER ASRAT ZEWDU ID: SGS/0279/2013A

> July, 2022 ADDIS ABABA ETHIOPIA

ASSESSMENT OF PROJECT MANAGEMENT PRACTICES IN HIGH-RISE BUILDING: THE CASE OF CBE HEADQUARTER BUILDING PROJECT

BY MAHIDER ASRAT ZEWDU

A Thesis submitted to the School of Graduate Studies of ST. Mary's University in partial fulfillment of the requirements for the Degree of Master of Arts in Project Management

> July, 2022 ADDIS ABABA ETHIOPIA

ST. MARY'S UNIVERSITY SCHOOL OF GRADUATE STUDIES DEPARTMENT OF PROJECT MANAGEMENT

ASSESSMENT OF PROJECT MANAGEMENT PRACTICES IN HIGH-RISE BUILDING: THE CASE OF CBE HEADQUARTER BUILDING PROJECT

BY

MAHIDER ASRAT ZEWDU

ADVISOR: YILKAL WASSIE (ASST. PROF.)

Dean, Graduate studies	Signature	Date
Advisor	Signature	Date
External Examiner	Signature	Date
Internal Examiner	Signature	Date

APPROVED BY BOARD OF EXAMINERS:

DECLARATION

I, Mahider Asrat, hereby declare that the thesis work entitled "Assessment of Project Management Practices in High-Rise Building: The Case of CBE Headquarter Building Project" submitted by me for the award of the Master of Art Degree in Project Management at ST. Mary's University, is original work and it hasn't been presented for the award of any other Degree, Diploma, Fellowship or other similar titles of any other university or institution.

Name

Signature and Date

St, Mary's University, Addis Ababa

July, 2022

ENDORSEMENT

This thesis is submitted to St. Mary's university, school of Graduate studies for examination with my approval as a university advisor.

Advisor

Signature and Date

St, Mary's University, Addis Ababa

July, 2022

ACKNOWLEDGEMENT

First and foremost, all praise goes to the Almighty God who helped me to accomplish this paper successfully. I am deeply indebted to many people for their contributions in diverse ways towards the successful completion of this thesis. Secondly, I pay my gratitude to my advisor, Yilkal Wassie (Asst. Prof.), for his comprehensive support, unreserved commitment, and constructive comments from the beginning to final stage of my work. Thirdly, I would like to take this opportunity to express my deepest regards and appreciation to my beloved family who had shown their holistic willingness to sponsor my academic carrier. Fourthly, I would like to extend my sincere gratitude to all participants and respondents of the study for their contribution. Finally, I thank all those who have helped me directly or indirectly in the successful completion of my research.

Table of Contents

Declaration	iv
Endorsement	v
Acknowledgement	vi
List of Figures and Tables	ix
Abbreviations and Acronyms	x
Abstract	xi
Chapter One	1
Introduction	1
1.1 Background of the study	1
1.2. Statement of the problem	3
1.3. Research Questions	6
1.3.2. Specific Research Questions	6
1.4 Objectives of the study	6
1.4.1 General objective of the study	6
1.4.2 Specific objectives of the study	6
1.5 Significance of the Study	7
1.6 Scope and Limitations of the Study	7
1.7 Organization of the study	7
Chapter Two	8
Literature Review	8
Introduction	8
2.1 Review of Theoretical Literatures	8
2.1.1. Construction Industry	8
2.1.2. High-rise Building	10
2.1.3 Definition of Project	12
2.1.4. Project Management	13
2.1.5. Projects management in construction industry	15
2.1.6. Project management challenges in construction industry	17
2.1.7. The High-rise building project development process	18
2.1.7.1. Project initiation	21
2.1.7.2 Project conception	22
2.1.7.3. Project realization/management	24

2.2. Review of Empirical Literatures	26
Chapter Three	31
Research Design and Methodology	31
Introduction	31
3.1. Research approach and design	31
3.3. Population and Sample Design	31
3.2.1. Target Population	31
3.2.2 Sampling method	32
3.2.3. Sample Size	32
3.3. Data collection tools and procedure	32
3.4. Methods of Data analyzing	33
3.5. Ethical Consideration	33
Chapter Four	35
Data Analysis, Results And Discussion	35
Introduction	35
4.1. The project management practice of CBE head quarter building project	35
4.1.1. Practice of the project conception process of the CBE New Head Quarter Project	36
4.1.2. The Design-Build (DB) project management practice of the CBE HQ Building project 4	40
4.1.3. Project Time, Cost, and Quality management Practices	46
4.1.3.1. Project Time management practice	46
4.1.3.2. Cost Management Practice	48
4.1.3.3. Project Quality management practice	49
4.2. Challenge in project management of CBE head quarter project	54
Chapter Five	58
Conclusions And Recommendations	58
Introduction	58
5.1. Summary of finding	58
5.2. Conclusion	50
5.3. Recommendation	51
References	53
Appendix I: Interview questions	70

List of Figures and Tables

Table 4.1: CBE head quarter project time performance	5
Figure 2-1: high-rise building project development process	. 20
Table 4.1: CBE head quarter project financial data	. 48

Abbreviations and Acronyms

AAU	Addis Ababa University	
AAiT	Addis Ababa Institute of Technology	
AEA	Association of Ethiopian Architects	
BoQ	Bill of Quantity	
CBE	Commercial Bank of Ethiopia	
CA	Contract Administrator	
CSCEC Ltd	China State Construction Engineering Corporation Ltd.	
СМ	Construction Management	
СРО	Construction Projects Office	
DB	Design and Build	
DBB	Design-Bid-Build	
FIDIC	Fédération Internationale des Ingénieurs-Conseils	
HQ	Head Quarter	
ICB	International Competitive Bid	
LDCs	Less Developed Countries	
NCB -	National Competitive Bid	
РМВоК	Project Management Body of Knowledge	

Abstract

High-rise building construction projects are very major part of construction industry which has an important role for growth of the nation. The challenges and risk factors in high-rise construction projects are high. Even though since recently high-rise building projects have been undertaken in Addis Ababa, very little studies were carried out on regarding the challenges and practice of managing high rise building projects. Accordingly, the study mainly aimed to assess the project management practices of the high-rise building construction in Addis Ababa with specific reference of CBE high-rise building project. The study used descriptive case study research design and qualitative research approach where field observation and semi-structure interview was used as primary source of data and documentation analysis as secondary source of data. The study used purposive sampling technique. Hence, the interview included one project manager, one contract administrator, and six project team members who were believed to know the area or subject matter very well. Different kinds of secondary data were also greatly used as triangulation mechanism. The qualitative research came up with a finding that DB project is becoming effective in meeting its project cost though there were some changes in cost were made due change in scope, however, the project is not effective in its time schedule as it has having significant time delay so far and the quality assurance is not certainly known though significant follow-up and supervision were made to ensure that project was in accordance with the details stated in the contract. The study identified the outbreak COVID-19, a lack of construction materials, a language barrier, a lack of certain norms and regulations, a shortage of qualified labor and contractor claims as major challenges of project. The study also identified that the project could have provided a significant opportunity for learning capacity and transferring technology and knowhow related with high-rise building projects. The study concludes that having skillful experts, documenting every project instruction in written form, providing discrete authority to the contractor with full responsibility (empowerment), harmonization of stakeholders, back-to-back evaluation with continuous progress review meetings were identified as how effective project term managed can be used to bring the project finally to be completed based on the experience of the case study.

Key words: High-rise building, Design-Build Project, Project, Project Management, Commercial Bank of Ethiopia (CBE)

CHAPTER ONE INTRODUCTION

1.1 Background of the study

Construction industry has substantial role in the advancement of the World economy. According to The Global Competitiveness Report (GCR), in 10 trillion USD revenue and added value of 3.6 trillion, the construction sector is account for about 13% of the world GDP. It accounts 7% of the world's working population (GCR, 2019). Its inter-sectoral linkage with other sectors, makes it to be powerful in the economy by having highly observable output and encourages sizeable amount of economic growth (Giang & Pheng 2010).

Particularly, in developing countries including Ethiopia, construction sector has very crucial contribution for rapid and equitable socio-economic development through output generation, employment creation, and income generation and re-distribution. According to Kerzner (2011), major construction activities in developing countries account for about 10% of their GDP and more than 50% of the wealth invested in fixed assets. The 2019/20 Annual Report by National Bank of Ethiopia depicted that the industry sector had 29.2% share in the GDP of the country; and Construction had a 68.5% share in the industry sector (NBE, 2019/20). However, despite this significance, the performance of the construction industry in the developing countries is generally low as indicated in many researches. The case in Ethiopia is no different. Researchers, such as Zelele (2016) and Yimam (2011), suggest that the level of construction project management practice in Ethiopia in terms of adapting general project management procedures, functions, tools and techniques is low.

Among the major problems facing construction project management today are those involving the managing of large construction project that involves the coordination of many diverse activities toward a common goal. In a large construction, project manager must devise plans which will tell with as much accuracy as possible how the efforts of the people representing different functions should be directed toward the project's completion. The major task of project management in such big construction projects required the coordination of various professionals working in the project, so that they have a maximal contribution to the project (Chartered Institute of Building, 2002). More so, According to Gorecki (2015), due to unique nature of a project, huge finance & resources required, diversified activities involved; the uncertainties usually come from such types of huge construction project. Improving project management capability has a role in reducing these uncertainties (Grant & Pennypacker, 2006). Here, assessing project management practice in general and such kind high-rise building construction project in general come of paramount importance.

High-rise building construction projects are very major part of construction industry which has an important role for growth of the nation. Construction of high-rise buildings had long been sought by human beings and has been actually realized. The realization and development of high-rise buildings has been possible due to the advancement of the engineering area such as the advent of steel, elevators and high-strength concrete along with the increased buying-power and human passion. The challenges and risk factors in high-rise construction projects are very high. Hence the various types of challenges involved in high-rise structures also play a major role in construction industry. Construction project objectives are always unique and built once. Various kinds of challenges all the time occurs at construction projects and frequently leads to time overruns or cost overruns or quality problem (Beedle, 2006).

Since recent time in Addis Ababa, buildings of 20 to 50 stories have been constructed these days and they are transforming the skylines of our capital city. In Ethiopia, a building greater between 35 to 150 meters, generally 12 to 50 stories, is considered as high-rise (MoUDC, 2019). Most of the high-rise buildings in Ethiopia are in the Addis Ababa. According to Emporis (2019), Addis Ababa currently has 101 Buildings which are considered to be high rise buildings that have total height of more than 45 Meter. In addition to these high-rise buildings being constructed, there are also more than 20 building under construction and new ones are being proposed and in progress. One of these is LaGare situated in the center of the city within proximity of the Addis Bole International Airport, with a rail line running along its northern edge. It has two tall residential towers, rising to their highest points of 34 stories in a "gateway" configuration at the end of the central spine (EHE, 2019)

The 48-storey CBE Tower is the tallest building in Addis Ababa, which is under construction near the Ethiopia Hotel off Ras Desta Damtew Street. The height of the building is 206 m (676ft)

and has a gross floor area of 165,400m² (1,780,351 ft²) that will serve as the headquarters of the state-owned bank. Early 2015, the Commercial Bank of Ethiopia signed a deal with China State Construction Engineering Corporation (CSCEC) for the construction of the building at a cost of \$267 million. The CBE Tower, which was designed by Henn Architekten, a German firm, was scheduled to get underway in 2011 but that never happened because the bank did not obtain approval from the State. Construction finally got underway in March 2015, now cladding work is being finalizing. On final completion, CBE Tower will be the tallest building in Ethiopia and it will feature prominently in the list of tallest building in Africa. Even though high-rise building the challenges and practice of managing high rise building projects. Accordingly, the study mainly aims to assess the project management practices of the high-rise building construction in Addis Ababa with specific reference of CBE high-rise building project.

1.2. Statement of the problem

Construction of high-rise buildings had long been sought by human beings and has been actually realized. Ordinary people have become more aware and interest in high-rise buildings due to the World Trade Center disaster which happened in 2001 in New York and a considerable number of people seem to have an ambiguous anxiety about the height of high-rise buildings. Nevertheless, the trend of constructing high-rise building is expected to continue. It is because not only high-rise buildings integrate all of the state-of-the-art construction technologies, but also they tend to be seen as a national symbol or indicator of economic growth (Beedle, 2006).

However, construction projects in general and high-rise building project in particular are somewhat difficult to manage and challenging due to the nature of the industry and the project; such as complex and unique nature, mobile workforce, ingrained culture, working conditions, and project-based setup, diverse sub-contractors and suppliers; in addition to regulatory bodies and changes in government policies during the production process (Arditi and Balci, 2009). These factors significantly affect the efficient performance of construction site management team (Fapohunda and Stephenson, 2010) and in most cases lead to project failure (Kar, 2009). The reasons for such failure are quite complex and it is not wise to pinpoint specific reasons to ensure project success. However, it is felt the main reasons for project failure in developing countries are lack of advance planning, a holistic approach, lack of comprehensive engineering and

management strategy, inconsistency in monitoring and follow-up, coordination and communication lapses and above all, absence of a methodical approach (Kar, 2009). Besides technical issues, posit that professionals in the construction industry also need to handle managerial issues (Agundu, et, al., 2003).

The situations are also common in the Ethiopian construction sector in Ethiopia in general and Addis Ababa in particular. Currently, lots of high-rise building projects have been constructed in Addis Ababa. In this regards some distinctive facts can be found in the start and development of Addis Ababa's high-rise buildings. Firstly, the technological know-how skills and experience in building and designing high rise building is at infant stage in country. This meant that the level of technologies required for super high-rise building projects was not fully understood. Secondly, there were not enough even there were also limitation execution and enforcement the existing laws and regulations which result in lack of disaster management system and social recognition at the time of developing super high-rise buildings. Consequently, it resulted in the loss of opportunity to construct land-mark buildings due to the lacking of faith (confidence) of concerned authorities and lack of effective regulatory framework. Furthermore, bureaucracy of relevant laws and regulations is still preventing the development of super high-rise buildings (Zelele, 2016).

Current super high-rise building projects in Addis Ababa have faced with several problems. Projects are being performed by a few construction companies of large size due to the characteristics of the construction-centered industry. This discourages small-sized design firms with an unstable financial foundation, from developing key construction technologies for super high-rise buildings. Related laws and regulations adhering to partition-style are not fully reformed and deep exclusionism exists among the general contractors, specialty contractors and consultants. This has also resulted in not only the obstruction of the application of advanced construction management (CM) methods to integrate design and construction, but also the joint research and development through the partnership of the companies. Consequently, the super high-rise building era has begun without any key items such as laws and regulations, core technologies, construction management technologies, and social consensus (Zelele, 2016). Each problem in the execution of super high-rise building projects in Addis Ababa possess various challenge which will require and entails further investigations.

Timely completion of a construction project is frequently seen as a major criterion of project success by clients, contractors and consultants alike. Newcombe et al. (1990) has noted that there has been universal criticism of the failure of the construction industry to deliver projects in a timely way. NEDO (1983) states that a disciplined management effort is needed to complete a construction project on time, and that this concerted management effort will help to control both costs and quality. This is tantamount to saying that the client's objectives can be achieved through a management effort that recognizes the interdependence of time, cost and quality. However, the CBE head quarter project had reported time delay. Due to the incorporation of additional element (scope) in project and unforeseen conditions associated with epidemics of Covid-19, the project had been extended for four times for a total of 813 days. The CBE head quarter project time performance is presented in table 4.1.

Date of Signature:	April 28, 2015
Contract Time:	1460 days plus 365 days given for defects
	liability period plus 10 years guarantee period
Mobilization Time:	90 Days from signing date
Commencement Date:	July 27, 2015
Completion date	July 23, 2019
Extension due to supplementary agreement	6 months
Amicable time extension No.1	10 month and 3 days
Amicable time extension No.2	180 days
Amicable time extension No.3	180 days
Revised completion date	January 30, 2022
The actual project inauguration date	February 13, 2022
The total project extension period	813 days
Time elapsed in Date:	2343 days
Time elapsed in %:	>100%

 Table 1.1: CBE head quarter project time performance

Source: project document (CBE, CPO)

More so, despite the fact that construction of high-rise buildings is under construction in our country, from the reviewed literature and to the best of the researcher knowledge there is no previous work on assessing the project management practice of High-rise building construction project. By assessing the project management practice of High-rise building construction project in the case CBE head quarter building project, this study fills the literature gap concerning high-rise building construction. This study also aims to identify various challenges which occur in the construction of high-rise buildings with specific focus on CBE head quarter building, key construction technologies and project management methods, which are integral for the construction of high-rise buildings. The study will also lay a foundation to further understand implementation of project management processes in other high-rise building strategic public projects in Ethiopia

1.3. Research Questions

The general research question that the study tries to answer was:

How was the project management practiced in the context of high-rise building project in Addis Ababa?

1.3.2. Specific Research Questions

The specific research questions that the study tries to answer were:

- How the project cost, time and quality of CBE head quarter building project managed?
- ♦ What are the challenges faced in managing CBE head quarter building project?
- ♦ What lessons can we learn from managing such high-rise building project?

1.4 Objectives of the study

1.4.1 General objective of the study

The general objective of this study was to assess the practices of managing high rise building in Addis Ababa with the case of CBE head quarter building project.

1.4.2 Specific objectives of the study

The Specific objectives of the study include the following:

- ⇒ To assess the project cost, time and quality management practice of CBE head quarter building project.
- \Rightarrow To identify challenges of project management practice in this project.
- \Rightarrow To suggest lessen that we learn in managing such kind of high-rise building project.

1.5 Significance of the Study

The findings of this research are expected to contribute a lot for different project stakeholders, project managers and project teams undertaking similar projects. First of all, the study will provide an important insight about the challenges of managing high-rise building construction projects. The finding of the study may also help the company manager to have better knowledge and understanding regarding the gaps in the current project management practice. More, it will also give a better insight about the mitigation measures that need to be taken to resolve those challenges regarding management of high-rise building. Finally, the study may serve as a reference for further studies

1.6 Scope and Limitations of the Study

The scope of this study was delimited to case study of CBE head quarter tower in Addis Ababa; which may restrict generalization of the findings to all high-rise building projects all over the city and the country. The conceptual scope of the study aims to describe and explain the project management practice with specific focus on project initiation, project conception and project realization (cost, time and quality) management practice of high-rise project in Addis Ababa. It specifically investigates how project management is being practiced in CBE head quarter building project in Addis Ababa by collecting data at a point in time.

1.7 Organization of the study

The study is organized into five chapters. Chapter one discussed the introduction part. It contains the background to the research study, presents the statement of problem, objectives significance and scope of the study. Chapter two contains theoretical review, empirical review of previous studies and conceptual framework of study. Chapter three outlines the research methodology adopted in this study. Chapter four discusses about the data analysis and interpretation of the outputs. Chapter five outlines the summary of the finding, conclusions, recommendations and further research suggestions.

CHAPTER TWO LITERATURE REVIEW

Introduction

This chapter includes the theoretical and empirical reviews which are found to be essential to the research inquiry. Thus, the first section discusses the theoretical literatures related to the study variables which are considered in order to lay solid foundation for the research. Besides, project management practice in the construction industry and related studies are also discusses in this chapter. In the final analysis, a brief summary of some of the related previous work on this study are discussed.

2.1 Review of Theoretical Literatures

In this section the conceptual definition of basic concepts concerning the main study variables are discussed. More so, the section also provides review of theoretical literatures concerning high rise building and other related issues of the study.

2.1.1. Construction Industry

The construction industry assumes a key part in deciding the advancement and prosperity of a country, and fundamentally adds to upgrading the stability and security of a nation (Ismail, et al., 2013). This could be attributed to the **fat** that the industry encompasses almost all aspects within the country be that related to irrigation, concrete structures, communication networks and all related civil engineering projects and initiatives. It therefore has wide applicability in such diverse sectors as health, education, agriculture, housing, commercial, utilities etc., all of which are related to infrastructural development and associated progress. It can therefore be concluded that the construction sector is an amalgamation of diverse organizations, impacting the construction process in varied ways.

According to Mulinge (2014), construction industry is an industry which is involved in the planning, execution and evaluation (Monitoring) of all types of civil works. Physical infrastructures such as Building, Communication and Energy related construction works; Water supply and Sewerage civil works, etc are some of the major projects / programs in the construction Industry. Leibing, (2001) identified the following basic features that are found in definitions of the construction industry:

- Massive: in the terms of the money that projects can cost the worldwide total is virtually uncountable
- Dominant: construction is a major factor in the national economy and contributes to the growth of geographic areas and employs a huge workforce.
- Flexible: the construction industry can adapt rapidly to any climate, situation, or need.
- Active: due to its continuous movement toward better quality, higher/newer technology, safer structures, and better services
- Structured: involves every feature of a project from financing to management to regulatory controls, to the actual construction design and execution (Leibing, 2001)

According to Chitkara (2011), a construction project is a high value, time bound, special construction mission of creating a construction facility or service, with predetermined performance objectives defined in terms of quality specification, completion time, budgeted cost and other specified constraints. According to Chitkara (2011), a construction project is a high value, time bound, special construction mission of creating a construction facility or service, with predetermined performance objectives defined in terms of quality specification, completion time, budgeted cost and other specified construction mission of creating a construction facility or service, with predetermined performance objectives defined in terms of quality specification, completion time, budgeted cost and other specified constraints. Construction can be considered as a dynamic industry which is constantly facing uncertainties, because building project is multidisciplinary which involved many parties as the project owner and various professionals, contractors and suppliers, manpower and subcontractors, these uncertainties and the many stakeholders in these kinds of projects make the management of costs, time and quality of project quietly difficult

Construction is a vast and an active economic sector, which is also considered as backbone of the world's economy in general (Panassaya, 2015). According to Global Construction Category Intelligence Report (GCCIR), in 2020, the global construction output value at 10.25 Trillion Dollar. It will also cause a marked rise in job availability due to the number of people the industry employs, which is about 7.6% of workers around the globe in 2020 (Global Data, 2020)

Particularly in Ethiopia, this sector mobilizes an enormous amount of various resources and budgets that embraces huge manpower of different professions by creating a large job opportunity (Lucy, 2016). According to the 2020 edition of African Economic Outlook, the market value of the construction sector in Ethiopia is currently estimated at more than US\$14.8

billion which accounted for 12.9% of GDP at current prices during the 2019/20 fiscal year. Global Data has cut Ethiopia's construction growth to 3.1%, from an earlier estimate of 7.8% in 2020, and to 5% in 2021 (Global Data, 2020).

2.1.2. High-rise Building

High-rise buildings are being built in connection with the growing population of cities, lack of land, as a rule, these are prestigious projects of special architectural significance, reflecting the state of the scientific and technical process, innovative technologies in construction. High-rise buildings have features that significantly distinguish them from ordinary buildings. In the following section brief review and discussion is presented as follows.

Various definitions are given to describe what a high rise building is. According to Craighead, (2009), any structure where the height can have a serious impact on evacuation is classified to be a high rise building. According to the building code of Hyderabad, a high-rise building is one with four floors or more, or 15 to 18 Meter (49 to 59 ft) or more in height. Emporis (2000), defines a high rise as a multistory structure between 35-100 meters tall or a building of unknown height from 12-39 floors while NFPA (2016), states that a high rise only comprises of buildings higher than 23 meters or about 7 stories. The *International Conference on Fire Safety in High-Rise Buildings* defined a high-rise as "any structure where the height can have a serious impact on evacuation". In the U.S., the National Fire Protection association defines a high-rise as being higher than 75 feet (23 m), or about seven stories.

In Ethiopia, provisions in 1986, 2002 and 2010 were made for building height classification. The new regulation for building height classifies building heights based on building height zones. The zones are elaborated as follows (AACPPO, 2017):

- Building Height Zone 1: -has 70m minimum height and adds no limit on the maximum height.
- Building Height Zone 2: -Maximum building height of 70m.
- Building Height Zone 3: -Maximum Building height is 35m.
- Building Height Zone 4: Maximum Building height is 35m but the floor area ratio is different from zone 3.

As we can see from the above definition there is universally accepted standard definition for high-rise building. It is defined differently in terms of height depending on the jurisdiction. Most building engineers, inspectors, architects and similar professionals define a high-rise as a building that is at least 120 feet tall. Considering this fact, this paper uses Building Height Zone 3 and above buildings which is similar to the Emporis (2019) standard definition of high rise. Therefore, the research evaluated and studied buildings with a height of more than 35 meters.

Throughout history, mankind has been striving for something more perfect, more ambitious, erecting ever higher and more refined buildings. The first high-rise buildings were built in the USA (Chicago). The increase in land prices, technical progress in building and construction and the reliable construction of the elevator by the American engineer E. Otis led to the appearance of the first many-story houses, which were called "skyscraper". At the turn of the 19th and 20th centuries, large volumes of office construction, as well as requirements for high concentration and building density, led to an increase in the number of storeys in buildings. At the same time, in addition to the possibility of erecting high-rise buildings on small land plots, the prestige and advertising of firms located on them played a significant role. That is why high-rise buildings are sometimes called "prestige buildings", and the architecture of these buildings is called "the big business style". n Europe, high-rise buildings appeared only in the middle of the 20th century. At first, these were detached buildings in a historic city area: Pirelli in Milan, Montparnasse Tower in Paris, etc. Since the 1960s high-rise buildings are located mainly outside urban historical zones. In the 1990s the most dynamically high-rise construction is developing in the countries of Southeast Asia (Saatcioglu, 2016).

The first high-rise buildings in Ethiopia were erected in the late 1950s -early 1960s in Addis Ababa. The earliest and most significant of which are 15 story Arada Building which has found around Piassa, Addis Ababa. According to Emporis (2019), Addis Ababa currently has 101 Buildings which are considered to be high rise buildings that have total height of more than 45 Meter. There are also more 20 high rise buildings which are currently under construction that have total height of more than 70 Meter.

According to Gould, et al, (2002), the high-rise building has posed some special challenges that are not usually encountered in other building construction. High-rise structures pose particular

design challenges for structural and geotechnical engineers, particularly if situated in a seismically active region or if the underlying soils have geotechnical risk factors such as high compressibility. They also pose serious challenges to firefighters during emergencies in high-rise structures. New and old building design, building systems like the building standpipe system, fire sprinkler system and other things like stairwell and elevator evacuations pose significant problems.

2.1.3 Definition of Project

First of all, it is essential to understand what is meant by 'project'. Many authors have defined projects in several ways. Here after going to discuss the various definition of project.

While there are several definitions of projects in the literature, one of the best has been offered by Tuman (1983), who states: "A project is an organization of people dedicated to a specific purpose or objective. Projects generally involve large, expensive, unique, or high-risk undertakings which have to be completed by a certain date, for a certain amount of money, with some expected level of performance. At a minimum, all projects need to have well defined objectives and sufficient resources to carry out all the required tasks."

In the words of Turner (1999), "a project is an endeavor in which human, financial and material resources are organized in a novel way to undertake a unique scope of work, of given specification, within constraints of cost and time, so as to achieve beneficial change defined by quantitative and qualitative objectives." The definition suggests three key targets of the project, i.e. time, cost and quality, which are to be in focus when undertaking the project. It also highlights the importance of efficient organization of available resources in order to achieve a good final result. Declerck et al., (1983, 1997), illustrate the political perspective of projects in this way: "a project is a whole of actions limited in time and space, inserted in, and in interaction with a politico-socio-economic environment, aimed at and tended towards a goal progressively redefined by the dialectic between the thought (the project plan) and the reality."

As defined in a Guide to the Project Management Body of Knowledge (PMI, 2013), a project is "a temporary endeavor undertaken to create a unique product or service." Temporary means that every project has a definite beginning and a definite end. The end is reached when the project's objectives have been achieved, or when it becomes clear that the project objectives will not or

cannot be met, or the need for the project no longer exists and the project is terminated. Unique means that the product or service is different in some distinguishing way from all other projects or services. Though the desired end results may have been achieved elsewhere, they are at least unique to the organization. Moreover, every project has some elements that are unique (Manalebih, 2018).

In lines of the definition provided by Pinto (2007), and accepted for the purpose of this research, a project can be defined as possessing the following characteristics:

- A defined beginning and end (specified time to completion)
- A specific, preordained goal or set of goals (performance expectations)
- A series of complex or interrelated activities
- A limited budget

So, the term project is very broad term and it is defined differently by different scholars. Even if they define it by their own understanding it represents the same idea. There is no one universally agreed definition of project. But it all definitions of scholars gave similarity. Accordingly, the definition of project shares some communality such as an endeavor activity, temporary or time bound, goal-oriented and that has unique deliverables. By summarizing the above given definitions, for the purpose of this study project is defined as a temporary endeavor that has unique output and carries out multiple activities in order to accomplish an organization's objective by exploiting its resources and having a definite starting and ending points as well as budget and quality constraints.

2.1.4. Project Management

From the above given definitions and characteristics of projects, it becomes very clear that projects require an appropriate form of project management practices that are different from the functional management approaches. Therefore, it is important to define the project management in general and construction project management in particular as the focus of this study is construction project, particularly, CBE head office building project.

Like project's definition, project management has also been defined in several ways. However, irrespective the nature of the project or the type of project in question, it is defined as the

management of project from its initial conception to its ultimate completion and its maintenance. In a typical project, the status changes from that of an idea or concept, through to feasibility studies, execution and final completion. According to PMI (2013), Project management is an application of knowledge, skills, tools and techniques to project activities to meet project requirements. This is accomplished through the application and integration of the project management processes of initiation, planning, executing, monitoring and controlling as well as closing (PMI, 2013). The PMI definition stresses the achievements of predetermined project objectives, which normally refer to scope, quality, time, cost and participant satisfaction, and directly links them to the project life cycle.

Traditionally project management was based on a belief that if projects are completed by the triple constraint of time, cost and performance, the project is assumed to be successful. However projects should also be viewed from an organization's strategic perspective for the purpose of providing value to both the customer and the enterprise as well (Polkovnikov et al. 2014). This is reflected in Wysocki (2014) definition who defined project management as "an organized common-sense approach that utilizes the appropriate client involvement in order to meet sponsor needs and deliver expected incremental business value". Here, business value is the responsibility of the client through their requirements statements. The project manager is responsible for meeting those requirements. Meeting requirements is the cause and incremental business value is the effect. Wysocki (2014) added to the above definition by introducing questions to which project management is designed to answer, these are: What business situation is being addressed by the project? What is to be done? How will it be done? How will you know you did it? How well did you do?

The Code of Practice for Project Management for Construction and Development (Chartered Institute of Building, 2003) describes the Project Management as an emergent professional discipline which separates the management functions of a project from the design and execution functions and defines Project Management as "the overall planning, coordination and control of a project from inception to completion aimed at meeting a client's requirements in order to produce a functionally and financially viable project that will be completed on time within authorized cost and to the required quality standards." This definition stresses the realization of

predetermined project objectives, which generally refer to cost, time, quality and a client's requirements throughout project lifecycle from inception to completion.

The common point of all the project management definitions is to consider project management as planning, coordination and control of a project along the whole life cycle of the project to meet the client's requirements. Besides, in lines of the definition provided by PMI, (2013) and accepted for the purpose of this research, managing a project includes: identifying requirements; addressing the various needs, expectations and concerns of stake holders; setting up, maintaining, and carrying out communications among stakeholders; managing stakeholders towards creating project deliverables and meeting project requirements; balancing the competing project constraints including scope, quality, schedule, budget, resources, and risks (PMI, 2013).

2.1.5. Projects Management in Construction Industry

The nature and characteristics of construction projects have some differences from other kinds of projects. The Construction extension to the PMBOK Guide by PMI (2016), described the generally accepted principles for construction projects that are not common to all general project types. Inherently, construction projects more often occur in an ever-changing, complex environment, and often with a high degree of risk.PMI (2016) further added that construction projects simultaneously address the geography, physical environments, existing infrastructure, communities, site conditions, and a wide range of stakeholder requirements. Moreover, PMI (2016) noted that construction projects typically carry large penalties or damages for projects that are completed late.

In construction, the term project management is frequently used to refer to site or construction management rather than taking a holistic view of the project from the conceptual stage (preparation of the client brief) to its ultimate completion and maintenance (facilities management). Walker (1984) provides a comprehensive definition for construction project management: "Construction Project Management is the planning, control and coordination of a project from conception to completion (including commissioning) on behalf of a client. It is concerned with the identification of the client's objectives in terms of utility, function, quality, time and cost, and the establishment of relationships between resources. The integration, monitoring and control of the contributions to the project and their output, and the evaluation and

selection of alternatives in pursuit of the client's satisfaction with the project outcome are fundamental aspects of Project Management" (Walker, 1984)

According to Kerzner (2003), construction management services are aligned with the activities and tasks associated with building design, construction documentation, construction procurement, and construction. The scope and approach to construction management is largely determined by the contractual arrangement established between the firm providing construction management services and the client. According to PMI (2004), managing a project includes:

- ⇒ Identifying requirements
- \Rightarrow Establishing clear and achievable objectives.
- \Rightarrow Balancing the competing demands for quality, scope, time and cost.
- Adapting the specifications, plans, and approach to the different concerns and expectations of the various stakeholders.

The major task of project management in construction projects is to coordinate various professionals working in the project, so that they have a maximal contribution to the project (Chartered Institute of Building, 2002). According to Hendrickson (2008), in addition to the knowledge of general and project management, the management of construction projects requires an understanding of the design and construction processes. Hendrickson (2008) summarized the functions of project management in construction as follows:

- Specifying project objectives and plans, this includes: defining the scope, preparing the budget and schedule, setting performance requirements, and selecting project participants.
- Maximum utilization of resources through labor, materials and equipment procurement in accordance with a plan and schedule.
- Implementing multitude of operations by properly coordinating and controlling the processes of planning, estimating, designing, contracting and construction.
- Developing effective communications and mechanisms for conflict resolutions among various stakeholders.

Overall, construction plays a major role in both economic and social development. The construction industry is the foundation for economic development. Construction is a crucial

sector for every society, economy, and culture, especially for developing countries (Knoepfel, 1992). Therefore, project management is vital in order to implement and control the construction projects efficiently.

2.1.6. Project Management Challenges in Construction Industry

Construction projects are somewhat difficult to manage and challenging due to the nature of the industry; such as complex and unique nature, mobile workforce, ingrained culture, working conditions, and project-based setup, diverse sub-contractors and suppliers (Arditi and Balci, 2009). These factors significantly affect the efficient performance of construction site management team (Fapohunda and Stephenson, 2010) and in most cases lead to project failure (Kar, 2009). The reasons for such failure are quite complex and it is not wise to pinpoint specific reasons to ensure project success. However, it is felt the main reasons for project failure in developing countries are lack of advance planning, a holistic approach, lack of comprehensive engineering and management strategy, inconsistency in monitoring and follow-up, coordination and communication lapses and above all, absence of a methodical approach (Kar, 2007). Besides technical issues, posit that professionals in the construction industry also need to handle managerial issues (Agundu, et. al., 2003).

Agundu, et. al. (2003), attest that various factor have been adduced for unhealthy scenario in management of construction projects in developing countries, the most notable being poor project analysis and management. However, Bowen et. al. (2002), argue that the concept of managing construction projects is deeply embedded in the traditional building procurement system. This is even as emerging project management. Kar (2009) contends that managing complex, multi-disciplinary projects in a developing country presents some special problems which vary from one project to another. However, Jacob, and McClelland (2001), argue that whether an organization manages stand-alone or multiple projects, whether those projects are small or large; most projects are difficult to manage because of two things:

- \Rightarrow They involve uncertainty, and
- \Rightarrow They involve three different and opposing commitments due date, budget, and content.

As the demand for highly innovative construction managers is constantly increasing, it has been acknowledged that management of construction projects from conception to disposal is difficult

and accompanied with enormous challenges. The construction managers deal with time, money, equipment, technology, people and materials in managing a construction project. They organize these resources into activities, execute the activities in logical sequences and manage to complete the projects within the stipulated time, budget and specified standard. The construction manager also manages the construction process to meet the clients' needs within legal, cost, time, quality and environmental constraints. In addition, construction manager takes the whole building cycle from inception to end of economic life, dealing with the procurement, construction, design or property management, recycling and disposal of building, and balancing the often conflicting requirements of clients, users and the community. Unfortunately, the complex nature of construction makes it one of the most adverse businesses that have ever existed. Thus, construction projects have often suffered from high fragmentation, large waste, poor productivity, cost and time overruns, and conflicts and disputes for a long time (Hai, et. al., 2012).

Moreover, Amalraj, et. al. (2007) agree that the challenges facing today's construction managers are bound to be formidable, while identifying quality, cost and schedule as some of the aspects of project management that are particularly challenging. Many of these challenges are a direct result of construction operations, while others a result of indirect, peripheral activities (Muir, 2005). Othman (2013) classifies challenges bedeviling mega construction projects in the developing countries into four categories of Engineering Challenges, Human Development Challenges, Managerial and Political Challenges and Sustainability Challenges. A surprising number of these challenges are not construction issues but must be addressed and managed by the construction manager to ensure project success. While some of the construction issues include workforce considerations, safety, time constraints, and the changing nature of the work, non-construction challenges that construction managers face that are part of the business landscape include legal issues, government regulations, environmental concerns, and socio-political pressures (Hai, et. al., 2012).

2.1.7. The High-rise building project development process

This section provides a thorough understanding of the nature and processes of building construction project in general and high-rise building project development in particular. Besides,

the emphasis of this section is on the presentation of potential challenges associated with the various stages of the process.

High-rise building project development is a highly complex, dynamic and multi-disciplinary endeavor, which would be well described in terms of its actual content by means of a process-related perspective. Millington (2000) notes with regard to the fundamental significance of the High-rise building project development process: "The development process is our most challenging manufacturing process because its sub-systems are complex and because it is the instrument of change which affects all of a community and a society" (Millington. 2000).

In the case of high-rise building project development, the process starts with the three factors of location, project idea and capital and ends with the building being ready for occupation. Various authors take different approaches, with differing degrees of detail, in mapping the high-rise building project development process by defining its individual phases (Wiegelmann, 2012). Flanagan and Norman (1993) emphasize two aspects of any construction project: the process, i.e., project phases, and the organization, i.e. project actors. From the process perspective, any construction project comprises a number of sequential phases. Different authors suggest a different number of project phases (PMI 2000). The simplest approach identifies two main phases – project development and project implementation. These two can be further detailed and developed into a larger number of phases, e.g. feasibility, design, procurement, construction, commissioning, and operation.

A PMI (2000), for example, propose a four-tier model, distinguishing between feasibility, planning & design, construction, turn over and set-up. Others such as (Harris et al. 2006) differentiate between eight phases, which are initiation, preliminary study, design and development, detail design, contract and procurement, manufacturing and construction, commissioning and operation and maintenance. Flanagan and Norman, (1993) identified the four phase of construction project which are investment decision, design, construction and occupy/use. Besides, Chapman and Ward (2003), differentiate between four phases, conceptualizing, planning, execution and termination. Despite the existence of different process models with varying numbers of phases, however such models generally cover mostly the same

tasks performed by a developer. Figure 2.1 overviews the different models presented in the literature.

For purposes of this study, the processes identified are consolidated into four main phases, namely: Project initiation, Project conception, Project realization/management and Project marketing /disposal. A similar phasing is presented by Gehner (2008). A generic flow diagram of the development process is presented in Figure 2-1. The figure illustrates how a project passes through the various stages of its development. Figure 2.2 the model used in the study.



Figure 2-1: high-rise building project development process

The model adopted in this research comprises only the first three phases – initiation, conception, project realization and management and exclude the last part- project usage/disposal. This is due

Source: adopted from (Gehner, 2008)

to the reason that the usage/disposal phase is excluded from the study because the project does not reach at the usage stage hence it is not important to consider it.

In summary, it should be noted that the complexity and dynamics of high-rise building project development process might be reflected in ideal-typical form by means of phase models. Nonetheless, it must be conceded that, in reality, the individual phases do not always take place in the sequence stated. In fact, real-life projects are generally characterized by overlaps, parallel operations and feedback effects, which cannot be mapped to a sufficient degree using phase models (Isaac et al, 2010).

2.1.7.1. Project initiation

The initiation phase commences the construction development process. A main expertise of a development organization is to identify the future demand on space market to create and provide an adequate supply and thereby to create value (Gehner, 2008). Creativity and drive are essential for a projects' success. Generating ideas within the framework of project initiation can, in principle, be divided into a level of factual analysis and secondly a level of inspiration and vision (Neary, 2009). Accordingly, for a development may either be:

- An existing plot of land, for which a use / project concept must be found and financing required;
- A project idea for which a suitable location must be procured respectively capital in search;
- The availability of capital seeking investment in a real estate project and thus a property/ micro location and project idea/project concept (Neary, 2009).

Main activities within the project initiation phase are commencing specific market re-search to ascertain demand from potential users/tenants and potential investor pro-files for the proposed development as well as preparing rudimentary development appraisals that will comprise the design, cost and program elements of the development. In case of unsatisfying outcome of the concept and its initial economics, the project will likely not be pursued any further (Gehner, 2008).

Based on a positive evaluation, the next major step is to typically obtain approval from the developer's senior management board and other significant stakeholders to proceed with the

initial concept. If the preliminary review is positive, the next step is to secure the required land in case the site is not already in the developer's possession or under exclusivity. In that case, a strategy for identifying and securing a site of suitable size, budget and location is to be elaborated. Often it is preferred by developers not to purchase the land at this stage but ensure exclusivity with the owner(s), given that a full feasibility analysis has not yet been completed (Gehner, 2008).

2.1.7.2 Project conception

The conception phase starts with the project feasibility analysis and ends in the implementation decision, or in abandoning the project. This phase can be qualified as one of the most important ones in the development process given its influence to the decision-making of the developer (Wilkinson & Reed, 2008).

Once the rough contours of the project have become visible in the preliminary acquisition review, what matters next is to outline the content of intellectual construct that was created in the initiation phase and to document it as a detailed project concept. This is ultimately intended to answer the question whether and in which manner the project is capable of being realized. According to Nozeman (2002) high-rise building concepts "comprise a great number of elements: function(s), location, size, branch (mix), target group(s), positioning, design, technical implementation/level of finishing, legal structure, marketing strategy, exploitation and management model." The term 'feasibility analyses has become accepted as a general term for the many types of analyses in advance of project implementation that are covered in this phase.

The goal of a feasibility study is to articulate a finding about the economic sustainability (feasibility) of the project under review. "A building project is 'feasible' when the construction project analyst determines that there is a reasonable likelihood of satisfying explicit objectives when a selected course of action is tested for fit of a context of specific constraints and limited resources" (Barkley, 2004). Prior to committing funds to a development project, a developer as well as his stake-holders and financing partners need a confirmation that market fundamentals will support the values assumed in the project appraisal (Barkham, 1997).

In terms of content, the feasibility analysis is based on detailed market and location analyses, building code reviews, design studies, use analyses, risk assessments, competitive analyses as well as profitability calculations. The challenge at this early and uncertain phase of the project is finding a balance between costs (potentially sunk costs), project uncertainty and the necessary quality and detailed specification of the usage concept (Wiegelmann, 2012).

i) Location analysis

The analysis of the location should critically verify the findings of the inception phase as documented in the preliminary acquisition review. The objective must be to obtain verifiable data that can be analyzed and presented in a manner to demonstrate to third parties the planned use of the land. These analyses are concerned with the long term-effective characteristics of micro- and macro locations. The location factors are both easily quantifiable "hard" criteria, as well as more 'soft' criteria, which will always retain some level of subjectivity (Wiegelmann, 2012).

ii) Project concept analysis

The building or usage concept for the use of the property must be based on the market and location analyses (micro and macro) discussed above. It examines the architectural and technical design of the building. Important criteria are the standard of specifications and the flexibility of the use of the building and its space efficiency. The objective is to meet market demand while minimizing cost (to build and operate) and maximizing flexibility (ibid).

iii) Competition analysis

The three above aspects of market, location and usage concept typically run parallel and are combined as the basis of a competition analysis, comparing the market position of the evaluated project with properties, which are or will be in direct competition. The first stage is the identification of appropriate benchmark properties. The objectives are to meet client needs while differentiating the development as much as possible from competitors. However, the weighting of criteria will always retain an element of subjectivity, which leads to residual risk.

iv) Risk analysis

While risks are present at all stages of property development, the feasibility analysis offers the opportunity to analyze them at a preliminary stage and review their impact prior to commitment of capital, as well as documenting and trying to mitigate such identified risks during later

implementation. To some extent, the progress of a development project through the phases of development has a general impact on its risk levels.

In its early stages of the development process, the initiation phase is characterized by a high degree of uncertainty and, in particular, creative and complex search and analysis procedures. At the end of this phase, success potentials and competitive advantages of building construction projects are identified and the project fundamentals defined.

Having assembled the above data and analyzed it based on appropriate assumptions, the results need to be presented and the developer will make a decision whether to proceed with the project. Progressing the feasibility analysis and making the project more concrete involves more effort and cost than optimal in case the project. The risk of sunk costs is ever present, but the level of detail required before a decision can be made should be obtained at reasonable cost, both internal and external. In the frame-work of the project initiation, it is the objective to answer the question in which manner and in what time the factors location, project idea and capital can be combined against the background of the strategy concept in order to produce a property that is competitive and acceptable in macro-economic terms Miles, et. al. (1995), have described this relationship as follows: *"Land, labor, capital, management, and entrepreneurship are needed to transform an idea into reality."* In case the project concept phase did not indicate that the developer's business requirements and objectives could be met, the project will likely be aborted. In the case of a satisfying out-come and outlook, the phase of project realization / management will be entered (ibid).

2.1.7.3. Project realization/management

The confirmation of the project's potential for success by the feasibility analysis triggers the initiation of negotiation and decision in the realization phase. At this point, at the very latest, the other parties to the project enter into the development process. These include the property owners, architects and engineers, building authorities and other representatives of the public interest, construction contractors, financial institutions, user groups, special service providers to the construction industry (project managers, consultants, etc.) (Wiegelmann, 2012).

The acquisition is made in the project realization phase by means of a binding right of purchase or the actual acquisition of the property to be developed. Finalizing the purchase can present unexpected difficulties and changes compared to the feasibility study base case as time has passed and stakeholder expectations are evolving. The price offered and agreed should be within the forecast parameters. Legal documents should be subject to appropriate due diligence and mitigation of execution risks. General risks that can occur during this phase include title issues which may not be satisfactorily resolved, inability to reach agreement on purchase/sale terms or inability to achieve a favorable quality of purchase agreement, purchase/sale terms which are less favorable than market comparable, as well as after purchase/sale additional issues that should have been discovered during entitlement and due diligence process (Barkham, 2002).

Another goal of preparing a more detailed usage concept is the definition of an optimal user mix on the basis of the feasibility study, which typically already includes a preliminary usage concept. In the sequence of the development process, this phase of the work is either performed after the acquisition of the property and in the course of the project planning process or - in a case of adequate or guaranteed certainty relative to planning - already during the feasibility study. Questions of building functionality, flexibility, building efficiency and architectural design are discussed as part of the usage concept. Thus, a further core task in connection with this phase is the preparation of a planning, implementation and contracting concept (Barkham, 2002).

In this stage it is important to undertake three main project realization/management activities which includes: project design, purchasing and construction

i) Project design

The objectives of the project design should be to balance the requirements of the in-tended user (functionality) with construction costs and sustainable operating and facility management costs, the expertise of construction firms, planning requirements, engineering considerations and aesthetic preferences in order to produce a project-specific optimum design for the site. Detailed plans for land, structural and capital improvements have to be prepared and necessary permits and licenses obtained. A significant challenge during this step is that the project design does not meet market needs and results in lower than anticipated rents or sales proceeds. Also, the initial project design may not address all regulatory issues. Costs to comply with regulatory requirements may reduce projected margin or return (Wiegelmann, 2012).
ii) Procurement

One of the main procurement tasks of the building construction project is to obtain a building permit within the schedule and on the basis of the previously developed usage type. The usual challenge during this stage is that bids from vendors/contractors require more time and or money than originally anticipated in the feasibility study, and that satisfactory contractors cannot be identified. Vendor negotiations may result in substantial revisions to project design (ibid).

iii) Construction

The construction phase starts with the granting of the building permit and the aim is the completion of the project within the planned framework of schedules, costs and quality. Once all necessary permits have been obtained, the developer gives the orders to start work. The building construction developer retains a coordination and internal reporting function. The building owner's functions that cannot be delegated are performed within the context of corporate management. All construction, planning and consulting contracts are entered into, and project controlling/ project accounting tasks are performed in this con-text. There are further obligations to act as representative vis-à-vis all project participants and especially vis-à-vis the public during the entire development period, as well as the task of reporting to the principal/investor or the providers of outside capital (Barkham, 2002).

2.2. Review of Empirical Literatures

In this section of the study, different reviewed literature relevant to the study will be discussed. The literature reviewed shows that there have been researches done to assess various organizations' project management practice in Ethiopia. However, there are gaps in literature concerning project management practice of building construction in general and high-rise building in particular.

Primarily literatures related to the project management practice of the Ethiopian construction industry are reviewed. Accordingly, Yimam (2011) conducted a study on the Ethiopian construction industry entitled "project management maturity in the construction industry of developing countries (the case of Ethiopian contractors)". The study aimed to assess whether and to what extent the processes, practices & tools under each project management knowledge area are being applied by the Ethiopian contractors in managing their construction projects. The research used interviews and questionnaires as an instrument and distributed it to 18 selected

Ethiopian grade I contractors. Based on the responses from the survey and interview, the research concluded that project management process maturity and practice maturity of the contractors is found to be at a low level. Also, the knowledge areas of material, procurement, cost, time, financial and human resource management have shown comparatively high maturity compared with other knowledge areas, and the project management maturity of contractors with ISO certification is found to be higher than those that didn't get certified.

Another research by (Hailu, 2016) was done to identify which processes of project management are effective for attaining the success of a project by studying cases of a successful and failed project. The research was carried out on two major building construction projects. The study on the two projects used a survey questionnaire as an instrument of the research. The research pointed out that if 88% of those effective knowledge area processes like quality, cost, time & communication processes are applied, then the success of a project is inevitable. The study output has shown that the triple constraints like cost, quality, and time, and the communication processes of the subject groups are the most effective project management processes towards the project's success. Moreover, planning processes from the process groups are useful for achieving the success of projects. The study also concluded that significant numbers of projects in Ethiopia are under the failed category. So, it recommends project management processes like planning, time, quality, cost, and communication processes have to be given considerable attention during implementation of the project.

Ayalew (2016), conducted a study titled "Assessment on Performance and Challenges of Ethiopian Construction Industry" to reveal the construction industry's level of project management practice. The study involved a literature review & used a structured questionnaire as a research instrument. The research involved 69 professionals from significant stakeholders of the construction industry. According to the response of the professionals and reviewed literature, the study concluded that the level of construction project management practices in terms of adopting general project management procedures, functions, tools & techniques to be unsatisfactory. Moreover, the study found that the level of practice of variables, such as cost, safety, risk, and time management, is to be deficient compared to their predetermined or planned values.

Befkadu (2017) conducted a study entitled "The practice of project management in Ethiopian real estate industry and its contribution to project success: The case of selected company in Addis Ababa". The major aim of this study was to examine the application of project management practice by depending on the various problems. To this end, the study used a descriptive research design including both qualitative and quantitative methods. Data was collected using questionnaire and interview from real estate companies, owners and consultants. Based on the response, IBM SPSS Statistics 20 was used to analyze the data. The major finding involves that project integration, scope, time, HR, procurement, and claim management are well managed in the Industry as well as Project initiation process groups and project closing process group are practiced well and consistently. However, from the knowledge areas project cost management, project quality management, project communication management, project stakeholder management and project risk management are poorly practiced in the real estate industry in Ethiopia. Therefore, it was recommended by the researcher to improve project management practice within the industry in Ethiopia (Befkadu, 2017).

Form the above studies, and one can infer that the maturity of project management practice in the construction industry is low, and the emphasis is not given to the major constraints of projects such as time, cost, and quality. Considering this, this study gives a particular emphasis on these areas of project management.

Secondly, literatures related to the project management practice of government and private organizations in Ethiopia are reviewed to understand the level of practice in different organizations. Accordingly, Karlsson (2011), conducted study entitled "the project management practice in Sweden & Ethiopia and potential improvements in project management methods". The research was performed on one of the most prominent private organizations in Ethiopia, Midroc, and its subsidiary in Sweden. It aimed to identify successful project management methods and share between the two organizations to increase efficiency and minimize the risk of construction projects. The research used observation, informal interview, and document analysis to come up with the result. The study concluded that there are several areas of improvement in Midroc Ethiopia and only a few in the Sweden counterpart. In the Ethiopian company, the planning process is not prioritized, and many of the projects start execution without sufficient planning. Also, project control is not prioritized, and rather than controlling; problems are solved

as they occur. Moreover, issues also arise in project management systems, integration management, human resource, cost, time, and procurement management.

Another study was conducted by Temesgen (2013), entitled "Assessment of project management capability: A case study at Mesfin Industrial Engineering PLC" set out to assess the company's current level of project management practice. The study used an interview & semi-structured questionnaire to collect data from the different departments of the organization. The research suggested that the company was operating at a foundation level and also that this maturity was not achieved by all the project management processes defined & planned during the planning stage is not consistently applied to all projects implemented by the company (Temesgen, 2013).

A study done on prepaid energy metering project in Ethiopian Electric Utility by Eden Tamru in 2018 had focused its topic area on assessing the project management practices particularly focusing on the life cycle of projects. Depending on a detail analysis made by the researcher using SPSS, it was indicated that a moderate level of project management practice meaning that various gaps were exhibited. The researcher reached at this conclusion by using a purposive sampling technique from the target population and used both qualitative and quantitative methods by gathering data from primary and secondary source. Descriptive research design and deductive research approach was used. Depending on the finding the researcher has recommended to develop the project management practices in accordance with the real practices on ground (Getachew, 2018).

Another research reviewed on assessment on the project management practice of Ethiopian Petroleum Supply Enterprise (EPSE) application and Network Infrastructure Project done by Behailu Negera in 2018. In relation to project process activities, the researcher tried to assess the practice of project management in EPSE depending Prince2 process groups. To this end, the study used a descriptive research design and both qualitative and quantitative methods by which data is collected in a form of closed questionnaire as well as secondary data source. By using SPSS software version 20.0, to analyze the responses gathered from the entire population. The researcher observed that gaps existed in quality assurance, risk and issue management and lessons learned from former projects (Negera, 2018).

From above research discussed, one can point out that the project management practice at the organizational level is also low as the construction industry. Although there are some practices adopted and some level of understanding of project management, there is a long way to go in implementing the practices properly. From the literature and empirical reviews, it can be concluded that adequate research had not been done in evaluating project management practices of the building construction project in Ethiopia.

From the reviewed literature and to the best of the researcher knowledge there is no previous work on assessing the project management practice of High-rise building construction project. By assessing the project management practice of High-rise building construction project in the case CBE head quarter building project, this study will fill the literature gap concerning building construction building. The study will also lay a foundation to further understand implementation of project management processes in other high-rise building strategic public projects in Ethiopia.

CHAPTER THREE RESEARCH DESIGN AND METHODOLOGY

Introduction

In this chapter, the researcher describes the procedures to ensure a methodical and well-informed investigation, focusing on sampling procedure, data collection and analysis methods. Data collection instruments and procedures are discussed as well as the target population and sampling procedures.

3.1. Research Approach and Design

Based on the purpose of the research, the researcher was utilized qualitative case study approach to assess the challenges of project management practices in the case of high-rise building project. The qualitative case study research approach is considered to be the most appropriate for this study for the reason that it allows the researcher to understand deeply and fully the phenomena and the problem under investigation. More so, the researcher was selected it because of the fact that it is suitable to investigate processes and challenges in addressing individual case problems, in this case the challenges of managing CBE head quarter project. It was therefore justified in view of the above descriptions and strengths that qualitative case study is the most suited and appropriate research approach for this study.

Regarding the research design, the study was utilized descriptive research design. The descriptive research design was selected for the reason that it helps to show the existing reality of real-world problems and provide essential fact about it. In this method the researcher was collected relevant data from the participants which were discussed in detail based on the objective of the study. In doing so, the qualitative data were used to assess the challenges of managing high rise building and to search a solution for it.

3.2. Population and Sample Design

3.2.1. Target Population

Target population is specified groups of people or object for which questions can be asked or observation are made to develop the required data structures and information. The major stakeholders who are actively participating in the project implementation are the very population of the study. This includes those staff members who are working in Construction Project Office

(CPO) of CBE and those who are working in Employee's representative side, which is Addis Ababa Institute of Technology (AAiT). The total numbers of professionals engaged in CPO are 15 including the Manager of the CPO of CBE and there are about 20 key professionals from the employee's representative side engaged in this specific project. This are mainly included project manager, project coordinators, project team members and supporting staffs who are directly involved in the managing and supervising CBE head quarter project in Addis Ababa.

3.2.2 Sampling Method

Regarding the sampling method, the study was applied purposive sampling. Purposive sampling technique is preferred for the reason that it allowed researcher to select the sample that fit the criteria and interest of the study. Moreover, purposive sampling was selected for the reason that it increases the relevance of the sample to the population of interest, as only individuals that fit particular criteria are included in the sample. This method also allows researcher to select respondents that have better insight, knowledge and experience on the issues and problems under investigations.

3.2.3. Sample Size

The sample size of the research is determined by purposively selecting respondents who are project team members who know the area or subject matter very well. In the interview the researcher was tried to include all appropriate individuals and project team members as much as possible in order to make the finding more robust. Meanwhile, the stakeholders namely from the Employer side; CBE CPO Project Manager, the Construction Lawyer and three engineers who were engaged in the project was selected for the interview based on research tools which were designed for each. From employer's representative sides, two site and office engineers who were undertaking the project follow up were parts in the interview.

3.3. Data Collection Tools and Procedures

The study was employed both primary and secondary sources of data in order to get appropriate data. As the study adopt qualitative case study approach, the researcher was mainly use semi-structured interview as a primary data collection tool which is helpful in answering questions related to the study objectives. To this end, semi-structured interview was prepared to probe

views and opinions of some purposively selected respondents. The interview was prepared based on the review of related literature important to the subject of the study. In this method, the participants, who are the practitioners in their field, were pass on their knowledge to the researcher through the conversations held during the interview process. It is selected for the reason that it offers the merit of using a list of predetermined questions as in a structured interview, while keeping enough flexibility to enable the interviewee to talk freely about any topic raised during the interview. In addition, data was collected through various sources such as relevant books, project contract, project management plan, project progress reports, articles and journals.

3.4. Methods of Data analyzing

Qualitative data analysis is an activity of "working with data, organizing it, breaking it into manageable units, synthesizing it, searching for patterns, discovering what is important and what is to be learned, and deciding what you will tell others" (Bogdan and Biklen, 1982 in Simon, 2011). Data analysis in qualitative research is an ongoing process that is undertaken concurrently with data collection, interpretation and report writing (Creswell, 2009). Whereas there are rules how to analyze quantitative data, there are no such explicit rules for qualitative ones (Bryman A. & Bell, 2011). Data analysis in qualitative research is an ongoing process that is undertaken concurrently with data collection, interpretation and report writing (Creswell, 2009). Considering this fact, in this study, the qualitative data that was obtained primarily through semi structure interview was subjected to in-depth scrutiny using thematic analysis. In this approach the data that were collected through semi structured interview would be analyzed by identifying patterns and relationships through scanning the interview information and then combining and summarizing the results accordingly. The researcher closely examines the data to identify common themes – topics, ideas and patterns of meaning that come up repeatedly. The thematic approach involves a six-step process: familiarization, coding, generating themes, reviewing themes, defining and naming themes, and writing up.

3.5. Ethical Consideration

The researcher did her best to address ethical consideration of confidentiality, privacy, and informed consent. All the study participants were informed about the purpose of the study and verbal consent of all study subjects were obtained before data collection. Participants were also

be informed that they have full right to discontinue or refuse to participate in the study. To ensure confidentiality, the name of interviewee was not be written on the questionnaire.

CHAPTER FOUR DATA ANALYSIS, RESULT AND DISCUSSION

Introduction

This chapter covers the presentation, analysis, and interpretation of data gathered through semistructured interviews with the project manager and members of the project team. To analyze the collected data in line with the overall objective of the study, qualitative analysis was applied by combining and summarizing the results.

The data contains key elements in major project management practices and major challenges faced in building of high rise building with the specific case of CBE head quarter building project in Addis Ababa which are drawn from interview data and different literatures. The data were collected from semi-structural interview and through reviewing secondary data. After the semi-structural interview was conducted and answers were obtained, qualitative data analysis method was utilized by means of the thematic approach to present the finding of the interview. The reliability of the reply of the interviewees also crosses checked through reviewing secondary data such as project plan, project contract, project status reports and other project related documents. Hence, this section provides analysis of the interview data. The discussion is also supported by findings from similar studies.

4.1. The project management practice of CBE head quarter building project

This section presents information on the project management practice of the CBE head quarter project. The researcher has made a semi-structured interview with the CPO Manager of CBE, the Construction Lawyer and three well experienced project officer and engineers of the project and two professionals from employer's representatives at their work premises. The interview questions raised were mainly about the project contract administration, the project cost, time and quality management practices and the majors challenged and lesson learn from under construction project of the New CBE HQ DB project and the result were presented as follows.

4.1.1. Practice of the project conception process of the CBE New Head Quarter Project

First and foremost, the study seeks to evaluate the CBE New Head Quarter Project's inception and conceptualization process. The study gathered data from semi-structural interviews and secondary data reviews to analyze the practice of project commencement and conceptualization, and the results are presented below. Before going into the practice of project initiation and conceptualization, it's vital to first go over the parties involved in a Construction Contract. Design - Build (DB) project delivery through FIDIC IV form of construction contract was used for the CBE headquarter building project. In major construction project, mostly there are three major parties: the employer, the consultant (Engineer) and the contractor.

The Employer (sometimes called the promoter) in this case Commercial Bank of Ethiopia, is the party who initiates pay for and is the ultimate owner of the project. The employer's role is as the initiator of the project and the end-user of the completed works. The main responsibilities of Employer are to appoint the Engineer (consultant) to select the contractor, to ensure that the site is available for the contractor, to pay the contractor and to agree any changes to the contract that becomes necessary. The FIDIC IV form of construction contract was used for the CBE headquarter building project, which allow the employer to limit the power of the Engineer (Consultant) and require the consultant) is not performing the duties properly, the employer has the power under the contract to replace the engineer (Consultant). In this case the Ethiopia Commercial Bank is the employer.

The contractor is the second party and whose main responsibility is to execute and complete the works and remedy any defects therein in accordance with the construction contract requires the contractor to undertake any of the requirements of the contract to the satisfaction of the engineer (Consultant). The contractor is obliged to provide all superintendence, labor, materials, etc., necessary to carry out and complete the project. The contractor is fully responsible for site operations and methods of construction. If the construction contract requires the contractor to undertake any of the design, the contractor is responsible for that design even if the engineer (Consultant) approves it. In this case, the well-known Chinese government owned Construction Company; China State Construction Engineering Corporation Ltd is contractor.

The Engineer (Consultant) is the third party in most forms of construction contract, who is appointed and paid by the employer. When a consulting firm is selected, the Engineer will act as a supervisor for the construction phase; possibly the firm may also have been the designer of the project. In CBE head quarter project case, there is not consultant. However, following the contact agreement signed with the contractor, the Commercial Bank of Ethiopia in the year 2016 has awarded the Addis Ababa University; Addis Ababa Institute of Technology (AAU-AAiT) for the Design Review, Supervision and Contract Administration of this CBE New Headquarter Design-Build Project. It is also important here that Addis Ababa Institute of Technology (AAiT) is not consultant it is rather advisor (representative) appointed by the employer to carry out certain duties on his behalf.

It was based on this that an agreement was made in April, 2015 between CBE and the wellknown Chinese government owned construction company; China State Construction Engineering Corporation Ltd. whereas, this company to Design and build (DB) the CBE New Head Quarter project and the bank has accepted an offer by the Contractor through the Letter of Acceptance in March 2015 and in April 2015 for the Design, Build and completion of such works and the remedying of any defects therein with.

Following the land acquisition on lease basis of the CBE new HQ project, a consultant was commissioned to prepare design program for the Architectural design competition. "Architectural Design Competition" was selected aiming to get "state-of-the-art" design for the CBE Head Quarter Building. After consulting the Association of Ethiopian Architects (AEA), National Competitive Bid (NCB) open to "association of local and expatriate consultants" was selected. The Bid for Architectural Design Competition is advertised in May 2009. The evaluation and selection of prizewinners was concluded and prize award ceremony was held on 29th of December 2009. The 1st prize winner was a company from Germany with a local Consultant of Ethiopia. While designs made by another Netherlands company in collaboration with another Consultant from U.A.E and an Ethiopian Consult in collaboration with the U.A.E consultant secured second and third places respectively.

Following the award ceremony, negotiation was held with the first prize winner. However, the negotiation failed due to disagreement between the local and foreign consultants and the

relatively high offer compared to the international fee rate. In spite of all these challenges, CBE through its Technical Advisory Group (TAG) has tried its best for the last to engage the partnership, though not successful. In conclusion, the negotiation was terminated after 17 months due to the reason that the two consultants are not good partners. The result made CBE to cost so much time.

After the failure of the Architectural Design Competition in 2009, CBE studied Alternative Project Delivery Methods and chose Design-Build as alternative project delivery system. Design-Build (DB) is an innovative method a Single Contractor (one agreement) undertakes Design and Construction of the project. The Selection of Project Delivery Method was made after recommendation by Technical Advisory Group (TAG) established for this project and endorsement by the Top Management of the Bank. Accordingly, the "Design-Build" Bid was floated from 15th September 2011 to 13th January 2012. After technical evaluation, three firms were technically qualified and passed for the next financial evaluation. Then, a negotiation was undertaken with the least evaluated bidders, until June 2012, which was discontinued before conclusion of Contract due to various internal and extraneous reasons.

The second-round negotiation was undertaken from October 2014 to April 2015 for a period of 7 months. During this period, CBE proposed additional Employer's requirement to make the building modern and more useful to the Bank for the next 50 years. Accordingly, the capacity of main hall is raised from 1500 seats to 2,000 seats. The total number of mini conference halls was raised to 7 (five halls of 200 and two halls of 300 seats). The total parking area is also increased from 1011 to 1500 stalls. At the conclusion of the 7 months strenuous negotiation on technical and financial matters, agreement was reached and it is approved by the Board of Directors of the Bank. Then the Contract for Design–Build of CBE New Head Quarter is signed with the China State Construction Engineering Corporation Ltd. on April 28, 2015. It is also important here to indicate that following this contact agreement signed with the contractor, the Commercial Bank of Ethiopia in the year 2016 has awarded the Addis Ababa University; Addis Ababa Institute of Technology (AAU-AAiT) for the Design Review, Supervision and Contract Administration of this CBE New Headquarter Design-Build Project. It is also important here that Addis Ababa Institute of Technology (AAiT) is not consultant it is rather representative appointed by the employer to carry out certain duties on his behalf. Here it is important to take one lesson that

unlike most construction project which followed Design-Bid-Build (DBB) approach, the CBE head quarter project is mainly followed the Design - Build (DB) contracts which is a new type of building project in the country. So far, the design-build (DB) was commonly practiced in hydropower and road projects.

There are three dominant delivery methods for a construction product: design-bid-build (DBB), design and build (DB) and construction management (CM) (Joyce and Gould, 2009; Winch, 2010). Design and build refer to the form of contracting where one supplier is responsible for parts of or the entire design and production (Winch, 2010). Design and Build (DB) is also known as turnkey contracting or single-point responsibility. Design-Bid-Build (DBB), also known as "hard bid" or "traditional method", on the other hand, is a type of project delivery system where the owner holds two separate contracts, one with the designer and another with the contractor. The designer assists the owner in developing the program and is responsible for design and the development of drawings and specifications. (Nicholas, 2010). DBB is the most practiced type of delivery system in the Construction Industry of Ethiopia since the year 1987. After project owners did prepare the basic planning that identifies construction project programs, they call upon the participation of Design and/or Supervision Consultants either by tender or by negotiated contracts. This consultant will carry out the design together with the necessary tender documents which will be the bases for tendering to select contractors. This process is called Design - Bid - Build and hence, the name for such delivery system.

The conditions of the contract signed between CBE and the contractor was based on FIDIC manual; Silver book- conditions of contract for Turnkey Projects (First Edition 1999). As the per interview response this edition was selected because it is suitable for use on process where, the contractor will take on full responsibility for the design and execution of a project. Risks for completion of time, cost and quality are transferred to the contractor and so the silver book was the only suitable for use with this highly experienced contractor with the ability to manage risk. Similarly, the contractors also assume responsibility subject to some exceptions, for the accuracy of the employer's requirements which is a major difference to usual design and build contracts.

The CBE has its own Construction Projects Office (CPO) which administers all the bank's construction projects. Contract administration is one of the most important jobs of the CPO

related to construction projects and involves numerous tasks occurring before and after contract execution and work order issuance. All works are administered by the CPO in accordance with the contract specifications, terms and conditions, Ethiopian laws and regulations, and the bank's policy.

4.1.2. The Design-Build (DB) project management practice of the CBE HQ Building project

As per the interviewees reply and documentation review, it was observed that, from the five major project delivery methods in public and private construction projects, the contract type used for the CBE HQ project is the Design-Build (DB) delivery method. As per the interviewees' reply, the reasons for selecting this delivery method were; the Design Build or sometimes-called package deal or turnkey contract is an innovative method where the contractor takes full responsibility for the design and construction. The DB is suitable for large and complex projects like the CBE HQ building project; where early completion is a requirement.

Actually, it is a shift of procurement strategy from the previously intended and partly implemented Design-Bid-Build (DBB) for this specific project by way of international architectural design competition. Even though Design-Build is a new trial in building projects in Ethiopia, it is not a novel procurement strategy as it is already applied domestic hydropower and road projects. Most probably, this is a big innovation in the procurement of building projects. As interviews explained, CBE has a practice of DBB contract delivery system all the times before this new HQ building project; like for the new Lideta building, Megenagna Building, Sebeta and others. But now, it has entered a new DB contract with the well-known Chinese Government owned Construction Company; China State Construction and Engineering Corporation Ltd. for the new HQ building project of the bank where, this contractor is responsible for all parts of or the entire design and construction of the building.

As per the interviewees reply and documentation review, there are three major parties in the construction project's setting; the Contractor, Employer and the Consultant. The Employer (CBE) is the Client or Owner of the project & is the one who pays for the work. The role of the Employer is to express its project requirement (Employer's Requirement), launch the project, to conclude contracts with the Contractor and Consultant, supervise the performance, make major

decisions (like variations, extension of time, etc.), make payment and to accept the to be completed project. More so, as per the interviewees reply and documentation review, the Current practice of the CPO of CBE contract administration in different phases of the project's delivery process consists of; record keeping, document control, payment management, schedule and cost control, progress reporting, management of variations and change orders, management of claims and disputes; if any.

The total number of staff of Construction Project Office (CPO) of CBE reported is 20. whereas on the contractor's side, 88 Chinese employees both from the professional and skilled; and there are also about 40 key professionals from the employee's representative side engaged in this specific project. In addition, the contractor hired 370 local unskilled daily laborers. This means that, almost all the professionals and skilled manpower of the contractor company are foreigners from its own citizens.

In the case of CBE, the main basis of effective contract administration of CBE-CPO is having a clear description of rights and obligations. For the contract administration to be effective, the Counterpart Engineers shall base on the following main principles and commitments:

- a) Has to address the needs of the Bank outlined, in the terms of reference, specifications for works, or services,
- b) Ensure that the objectives of the project are achieved by taking the appropriate and timely action,
- c) Ensure that contracts are based on clear and accurate description of deliverables, clear rights and obligations of all parties involved in the contract execution. D) Monitor performance, collect information and identify steps that should be taken to measure actual contract achievement
- d) Proactive administration of contracts which should be based on close follow-up of implementation of the contracts and anticipating problems and responses before they impact on the contract performance
- e) Successful contract administration based on open and effective communication between all parties using appropriate channels, means and providing feedback to enhance contract

performance. These could be in the form of progress meetings, follow-up emails, calls, etc.

- f) Identifying potential contractual risks such as not meeting the deadlines, receipt of less than acceptable quality deliverables and alerting relevant officers.
- g) Implementing a system of monitoring and evaluating the performance of contractors, consultants, suppliers and other service providers

Interviewees were explained the advantage of DB project delivery method. As per the interviewees reply and documentation review, when compared to DBB procurement route used for previous building projects of CBE, the DB contract delivery system of the CBE HQ project has an advantages for the employer on that; single point responsibility- the contractor takes all responsibility for design and construct, reduced completion time - overlap of design and construction significantly reduces completion time, reduced overall cost - less number of variations and claims, few change orders, less claims and disputes, better security through bonds and warranty.

The DB is suitable for large and complex projects like the CBE HQ building project; where early completion is a requirement. In explaining the reason behind selecting of Turnkey Project delivery on of the interviewee narrated that:

"The bank chose Turnkey Project Delivery because it is appropriate for usage in situations when the contractor will assume complete responsibility for the project's design and execution. A turnkey project is one in which a single entity— a contractor—works with a project owner to complete all stages of a project, from detail engineering to construction, under a single contract. Because this strategy places duties on the contractor that would otherwise fall on the owner in a standard design-bid-build (DBB) delivery, it eliminates difficulties for the project owner. Turnkey delivery approaches, unlike the DBB approach, combine features of project design, financing, procurement, subcontracts, and construction into a single-point-of-accountability solution. Because the risks of completion in terms of time, cost, and quality are shifted to the contractor, the silver book was the only option for this highly experienced contractor who could manage risk".

Specifically, cost and time advantage of turnkey project was reported during the interviews of the project team members. In this regard one of the interviewees explained:

"The main benefit of a turnkey project is that it eliminates financial unknowns, prevents project cost overruns, and allows project owners to precisely calculate return on investment (ROI), as well as minimizes the amount of ambiguity surrounding overall project cost. As you know, the construction cost is unknown using traditional approaches until the project is developed and bid out. If a company with little or no construction experience is selected to design and engineer the project, they may not be able to do so within the budget. This can result in building bids coming in significantly higher than anticipated. This is most of the case in projects in our country. However, in the turnkey project is bid fixed cost, project owners are only responsible for the lump sum agreed upon during the contract negotiation. Thus, any increase in costs, such as surges in material costs, are already accounted for and are covered by the contractor. That's why the current CBE headquarters project only cost 3033.3 million dollars to build; otherwise, the price would be much higher".

The CPO further refers that, according to a study made by the Pennsylvania State University in collaboration with the Construction Industry Institute (CII), compared to the DBB method, the DB procurement route reduces total project cost by 6%, cost overrun by 5.2% and reduces time of completion by 33%. It further mentioned that, similarly, as per a study made for the US Federal Highway Administration in 2006 has revealed DB projects are 20.5% cheap and completed in 24.4% reduced time.

On average, managers of design-build projects surveyed in their study estimated that design build project delivery reduced the overall duration of their projects by 14 percent, reduced the 31 total cost of the projects by 3 percent, and maintained the same level of quality as compared to design-bid-build project delivery. On the other hand, other project survey results revealed that design-build project delivery, in comparison to design bid-build, had a mixed impact on project cost depending on the project type, complexity, and size. The surveyed design-build project managers further indicated that project delivery approach (i.e., design-build versus design-bidbuild) can be a contributing factor in controlling and potentially reducing project costs. However, project delivery approach was perceived to be less of a factor in affecting project cost than other characteristics of the project or its participants.

Interviewees were explained the disadvantage of DB project delivery method. The interviewees specifically mentioned that the employer has found as a disadvantage of this DB project delivery that, need for having a detailed, complete, and up to date employer's requirement (performance specification) at the beginning of the project; difficulty to assess and manage the quality of works as the specification of the project is not complete and difficulty to include changes. In this regard one of the interviews explained that:

"The Disadvantage of the DB delivery system is loss of control cost of tender and cost of risks. Since limited supervisory role by the employer representative is practiced; which is relatively flexible and makes the employer distanced from the whole process, the employer has little chance to understand what is developed and entertain variations in requirements implying loss of control".

On the contractors' side, in order to provide reasonable offer, their tender cost is higher than in the case for DBB delivery system. This is because they need to carryout acceptable design for project cost offers. Though it was not practiced often, employers who shared costs related to tendering are informed to get seriously considered offers. World Bank suggested a two staged procurement method based first on technical merit and followed by financial competition and not for more than six bidders. The increase in risk transferred onto the contractor will be counterbalanced by the increase in contract prices which can be taken to include these costs of risks.

Regarding the relationship between the client and the contractor is a contractual relationship based on FIDIC manual; Silver book, 1999 and it is a strong binding agreement made; whereas the relationship between the contractor and the consultant is only a working relationship; i.e., no contract is signed between them. Since the consultant is only a representative of the employer, any formal communication made between the contractor and the consultant is also made copied to the employer. Hence, there is a triangular relationship among the employer, the consultant and the contractor. The relationship between the consultant and the contractor is based on professional basis as stipulated in the contracts. Complains (e.g., delay in response, assignment of inspectors, etc.) occur at different times and stages of the work. However, such complains do not happen frequently and are easily managed by their respective representatives.

As interviewees were stated, practically, complaints are handled by negotiation between the Contractor and Employer/Consultant and nothing has gone to the Dispute Review Board or Arbitration so far. There were minor claims which were solved by discussion and negotiation before turning to disputes like; who owns facilities for employer's representative staff, payment for temporary water, electricity, detail finishing level and partitions of inter. However, the contract has Amicable Settlement, Dispute Adjudication Board and Arbitration as dispute settlement mechanism. The interests of the main stakeholders (Employer, Contractor, and Consultant) are generally in line and the teams work together to achieve the project objectives. In some cases, the contractor focuses in the schedule performance of the project, while AAiT's team focus on quality of works in addition to the schedule and cost requirements. The interviewees also reported that even though the design review, supervision and contract administration is outsourced to the consultant (AAiT), CBE CPO engineer staff regularly follow up and administer each and every phase of the project as per the terms and conditions set in the contract in collaboration with the AAiT.

The client issues to the contractor instructions which are necessary to perform the obligations under the contract. Each instruction is given in written form with respect to the contracting terms, in which the obligations are specified. In the cases of variation, adjustments are made. The interviewees here again agree that the written form of communication facilitates the administration of the contract which also leads to its success eventually.

The contractor is doing the design, execution and completing of the works in accordance with the contract, and made remedies on any defects in the works. When completed, the works are to be fitted for the purposes for which the works are intended as defined in the DB contract. The contractor is responsible for the adequacy, stability and safety of all site operations.

In addition, the contract payment is 65% in the form of foreign currency and the remaining 35% in local currency so that there will be foreign currency drainage for the country. On the other hand, the participants have raised one positive point on this issue that, the contractor came to this

project with its own money; i.e., it doesn't need local bank finance to undertake the project and this is an advantage for the country; whereas; if local contractors own this project, they must need huge amount of bank finance.

4.1.3. Project Time, Cost, and Quality management Practices

The concept of managing construction projects is deeply embedded in the building procurement system. Ireland, (1983) argues that; time, cost and quality are the principal feasible objectives of the client in any construction project. Although it is claimed that time, cost and quality are incorporated in the management of construction projects; research has shown that in fact a time cost bias exists. Considering this the study assesses the time, cost and quality management practice of the CBE head quarter project. Accordingly, various questions related to the project time, cost and quality management practice were asked and the result is presented in the following section.

4.1.3.1. Project Time management practice

Timely completion of a construction project is frequently seen as a major criterion of project success by clients, contractors and consultants alike. Newcombe et al. (1990) has noted that there has been universal criticism of the failure of the construction industry to deliver projects in a timely way. NEDO (1983) states that a disciplined management effort is needed to complete a construction project on time, and that this concerted management effort will help to control both costs and quality. This is tantamount to saying that the client's objectives can be achieved through a management effort that recognizes the interdependence of time, cost and quality.

In terms of time management, in BD building contract, the Contractor takes the design risks and as such Request for Information (RFI's) and associated changes are minimal. However, the time management will consider the effect of excusable events on the schedule of the project. The Contractor makes its utmost effort to finalize the project in the intended time duration in order to get its contract payment early. In addition, the overlap of design and construction significantly reduces the completion time and hence, this project has been going based on its master time schedule for long time; but due to the occurrence of various international circumstances like COVID-19, which largely affected the global the construction economy, the project has reported the time delay.

It was observed that, the project has reported the time delay. An agreement was made in April, 2015 between CBE and the well-known Chinese government owned construction company; China State Construction Engineering Corporation Ltd. The project was commenced at July 27, 2015 after 90 days from the signing of agreement. The project completion date at per initial agreement was July 27, 2019. However, the project period was extended for three times. Accordingly, based on the supplementary agreement the project was first extended for six months period. This six-month extension was made due to the incorporation of additional element (scope) in project. In terms of project scope, based on the client request, supplementary agreement was signed to add two additional floors on the main tower and one additional floor in the commercial and conference building. More so, additional elements for building support systems such as Heat Ventilation and Air conditioner (HVAC), Pressurized firefighting, Security Camera Television (CCTV), etc. are included in the project which result in the first six-month extension.

The project also has been amicably extended due to unforeseen conditions for a second time (for 10 months and three days) and for third time (180 days). The project was being extended for the fourth time for 150 day and the completion date as per the extended contractual agreement was January 31, 2022. Over all, the project is extended for fourth times for a total of 813 days. Until January 31, 2022, 2313 days have been already elapsed from the total time duration of the project i.e., the time elapsed Vs. project period is already passed 100%. The building has been completed 97.8% from the total construction work as per the to-date progress report of CPO of CBE in January, 2022. Although some finishing work remains, the project was inaugurated with the presence of Ethiopian Prime Minister Dr. Abiy Ahmed on February 13, 2022, to coincide with the CBE's 80th anniversary. Over all, the project has taken 6 years and 5 months to complete.

As explain during the interview, the three consecutive amicable time extensions were made due the various external conditions. These three consecutive project extensions have been extended due to the occurrence of various unforeseen external and international circumstances which were outside the control of the contractor. In particular, the global epidemic of corona virus (COVID-19) was one of the main reasons for project time delay.

4.1.3.2. Cost Management Practice

Clients have been increasingly concerned with the overall profitability of projects and the accountability of projects generally. Cost overruns, in association with project delays, are frequently identified as one of the principal factors leading to the high cost of construction (Charles and Andrew, 1990). Research to date has tended to focus on the technical aspects of managing costs on construction projects in the attainment of client objectives. Considering this the study asses the project cost management practices and the result is presented hereafter.

Regarding the case whether the employer can entertain any cost overrun request of the contractor more than the total contract price agreed in the contract, the respondents mentioned that, since the contractor's offer under the DB route is a lump sum offer, this offer is little affected by price fluctuation. Contract Price Breakdown annexed with contractor's offer may give glimpse of the contractors' pricing and could be used for pricing variation. However, the lump sum offer has precedence over this contract price analysis. Hence, the relative certainty of the contract price makes it a preferred method of procurement for the employer. This leads to extended processes in payment certification process, as the amount executed in a given period cannot be easily converted to value.

Main contract agreement USD:	231,739,130.43
Discount (rebate)%- none	
Main agreement after rebate USD	231,739,130.43
Supplementary Agreement (B):	27,836,452.17
Additional Work - None:	
Variation work USD (D)	4,317,076.67
Revised contract Agreement (A+B+C+D)	263,892,659,28
VAT 15%	39,583,598.89
Revised Contract Agreement with VAT	303,476,558.17

 Table 4.1: CBE head quarter project financial data

Source: project document (CBE, CPO)

Additionally, as the cost per item is not known, the valuation of change orders and variations are difficult. However, since most of the risk is transferred to the Contractor, it may inflate the tender price initially. This increase in risk transferred onto the contractor will be counterbalanced by the increase in contract prices which can be taken to include these costs of risks. Due to the nature of DB building contract agreement (lump sum) the main responsibility of cost management mainly laid on the Contractor.

Although scope and design changes after the approval are difficult to administer and changes will be expensive to implement in DB projects, based on the negotiation made between the employers and the contractors some changes were made on the project which leads to the increase in the project cost. Thus, based on the interest of the client, the addition in scope of the project was made under supplementary agreement. In this regard, two additional floors in main tower and one additional floor in both conference and commercial building were added in the project which results in scopes change in the project. More so, additional elements for building support systems such as Heat Ventilation and Air conditioner (HVAC), Pressurized firefighting, Security Camera Television (CCTV), etc. are included in the project. Because of this, the project was completed with total cost of USD 303,476,558.17 from its initial contract value of USD 231,739,130.43 million.

4.1.3.3. Project Quality management practice

To the client, quality may be defined as one of the components that contributes to "value for money" (Flanagan and Tate, 1997). Vincent and Joel (1995) define total quality management as: "...the integration of all functions and processes within an organization in order to achieve continuous improvement of the quality of goods and services. The goal is customer satisfaction." Furthermore, in order to achieve successful project quality management three separate drivers to quality management must be managed, namely: (i) Integration of the project team so as to have a single objective and a common culture; (ii) A customer focuses for the team thereby facilitating the provision of products and services that will meet the client's needs and iii) A process of continuous improvement in the management of the construction project. When these three components are successfully integrated, the project begin to realize significant, measurable and observable improvements in the attainment of the clients' objectives.

As the explained in the interview, in order to achieve the quality of the project, the project has set three quality milestones which are: CBE requirement, standards, and practices.

Client Requirement: The first of the essential features of a design and build contract is that the employer approaches a contractor with a set of requirements defining what the employer wants. Requirements are capabilities that a project must meet to satisfy a user's need to solve a problem. The client's needs can come from a number of sources including compliance to a standard or to legal regulations, a business need, a business problem, market need, competition, etc. Investment in requirement processes implemented from the start of the project life cycle pays off at the end. In this regard, the head of PMO stated:

"In managing the quality of project, the first of the essential features of a design and build contract is that the employer approaches a contractor with a set of requirements defining what the employer wants. The contractor responds with proposals, which include production as well as design work. The scale of design work included depends on the extent to which the employer has already commissioned such work from others".

Standards: In managing the quality of project, standards are one of the three quality milestones that helps to manage the quality. According to the interviewees' response, the various standards were set in contract and all the works were undertake according to standard set-in project agreement. In explaining how the project is undertaken according to the stipulated standards the interviewee from the consultant (AAiT) narrated the following:

"In line with the stipulations of the design-build contract between the contractor and CBE, the consultant (AAiT) was responsible for managing any variation or change order requests made by respective contractual parties. AAiT review such requests and analyze their effect on the overall design and direct and indirect effect on the construction activities. AAiT prepare an assessment and recommendation report for CBE's and the contractor's team. Upon approval of such requests, AAiT follow up on the proper and effective implementation of such changes." **Practices:** the project also applied the best practices in quality management. In this regards the interviewees explain that the involvement of all relevant stakeholder as an important practice. The interviewees also reported that even though the design review, supervision and contract administration were outsourced to the consultant (AAiT), all the participants regularly follow up and administer each and every phase of the project as per the terms and conditions set in the contract in collaboration with the AAiT. In explaining how the project quality is monitored in relation to the standards. One of the interviewees from CBE CPO stated the following:

"The architect of the CPO works in each design review process with the consultant before it is approved. The civil engineers supervise the quality of the construction materials by making it a sample test at technology faculty department. Generally, they make in depth supervision on the materials. This mega project was being intensively constructed 24 hrs a day and the participants also work on the project site even during night hours in a shift basis. The Contractor's work quality and technical ability was inspected by the Employer and Employer's Representative (Consultant) through Design Review, Material testing at site laboratory & third-party Laboratory (AAiT Lab., at Ethiopian Conformity Assessment Agency (ECAA) and inspection by teams. If any of the parties become aware of an error or defect of a technical nature in a document which was prepared for use in executing the works, the party promptly will notice to the other party of such error or defect. Then accordingly things are settled in harmony. The client issues to the contractor instructions which are necessary to perform the obligations under the contract. Each instruction is given in written form with respect to the contracting terms, in which the obligations are specified. In the cases of variation, adjustments are made".

The interview also mentioned that the project applied written communication and documentation as an important practice for quality control. In this regards the interview mentioned that quality metrics and measures were centrally documented. The interviewees here again agreed that the written form of communication facilitates the administration of the contract which also leads to its success eventually. Respondents also mentioned how they tried to institute a quality assurance system to demonstrate compliance with the requirements of the contract. In this regard, one on the interviewee stated that: "The system was in accordance with the details stated in the contract. The client is entitled to audit any aspect of the system through the supervisors and project managers. Details of all procedures and compliance documents were submitted to the client for information before each design and execution stage was commenced".

Another important practice in relation to project quality management is the application of track trends. Trend quality metrics and measures over time to provide a graphical representation of trend of the project's conformity to defend quality standards. In this regard, the interview responses revealed that in addition to the progress review meetings used to be held every month; the consultant reports the daily and weekly status of the project to the client. Concerning the application of track trends practice, one of the interviews narrated the following:

"In terms of the application of track trends practice, in addition to the progress review meetings used to be held every month; the consultant reports the daily and weekly status of the project to the client. The participants in the monthly progress review meeting could be from any concerned offices other than the usual client, contractor, supervisor and project manager's presence. Visual aids (such as, picture figures) and other elaborative materials were used while presenting the progress, strategically perspectives, total payment and financial monthly figures used to be presented in the meeting. These meetings grant the follow up of not just the quality but also the time and cost variables of the project".

More the interviewee also mentioned the practice of managing quality as iterative process. This means that the project applied the practices in which the project quality management is an ongoing, iterative process that is conducted throughout the project lifecycle. Concerning the application of practice of managing quality as iterative process, one of the interviews narrated the following:

"We managed quality based on the principle that the project quality management is an ongoing, iterative process that is conducted throughout the project lifecycle. Reporting also continues until the contractor completes all work which was known to be outstanding at the completion date stated in the Taking-Over Certificate for the works. Each report includes charts and detailed descriptions of progress, including each stage of design, contractor's documents, construction, testing; photographs showing the status of progress on site, percentage progress and the actual or expected dates of the rest of the project tasks, records of contractor's personnel and equipment and comparisons of actual and planned progress, with details of any events or circumstances which may jeopardize the execution of the project in accordance with the contract, and the measures being adopted to overcome delays".

More in regards to project quality management, the respondents were also mentioned the practices of quick feedback or quick act. This means that the project obtains quality feedback as quickly as possible to avoid scalation of potential quality issues. When errors, omissions, ambiguities, inconsistencies, inadequacies or other defects were found in the contractor's documents, the works were corrected at the contractor's cost, notwithstanding any consent or approval. Though these costs were very minimal that it didn't raise complications or serious disputes.

The other point discussed by the participants regarding the project quality management practice was; the contractor imports almost all the construction materials from aboard, mainly from China except few, like cement, sand and gravel because, it doesn't believe that other local construction materials have high quality even though these local materials can be further attested by Quality Standards Authority; then local material suppliers could get additional market by supplying to this project. The contractor had also employed professionals for the project from China up to the skilled laborers. Only the unskilled laborers were hired from local people. i.e., key professional works were owned by the Chinese.

The interviewees also mentioned that administration of quality in this DB project was probably found to be more difficult when compared with that of the CBE DBB projects before. In this regard one of the interviewees stated:

"In this project, the complete specification and construction methods were not fully presented; i.e., all the interior work of the building was designed phase by phase (on-going design) during the project's implementation time and no Bill of Quantity (BoQ) was prepared at the beginning so as to know the exact specification and type of the materials to be utilized in the building project. This leads the consultant to assign additional inspectors to ensure the quality of works. However, the designer may not be under the direct control of the Employer because; usually the designer was employed by the contractor directly".

On average, managers of design-build projects surveyed in their study estimated that design build project delivery reduced the overall duration of their projects by 14 percent, reduced the total cost of the projects by 3 percent, and maintained the same level of quality as compared to design-bid-build project delivery. On the other hand, other project survey results revealed that design-build project delivery, in comparison to design bid-build, had a mixed impact on project cost depending on the project type, complexity, and size. The surveyed design-build project managers further indicated that project delivery approach (i.e., design-build versus design-bidbuild) can be a contributing factor in controlling and potentially reducing project costs. However, project delivery approach was perceived to be less of a factor in affecting project cost than other characteristics of the project or its participants.

4.2. Challenge in project management of CBE head quarter project

One major objective of this study was to identify challenges of project management practice in building CBE head quarter project. The following section discusses the major analysis and interpretation from the interview data regarding to the major challenges of project management in this project. All the interviewees highlighted several challenges. Various and diversified challenges were pointed out by the respondents. The challenges ranged from institutional, international, cultural, behavioral, conflict of interest, to environmental.

COVID-19: All the interviewees mentioned the outbreak of COVID-19, which they identified as a major challenge in the construction of the project. It is well known that the outbreak of Corona Virus in the world had a devastating effect on the world economy. The construction sector, in particular, was one of the worst hits. In relation to the challenges of COVID-19, one of interviewees stated:

"The outbreak of COVID-19 is the major challenge in this project. Related to this was the fact that the disease was first detected in China. The problem was exacerbated by the fact that most of the construction materials were imported from China. The occurrence of the corona virus was unpredictable and unforeseen condition and the impact was so great that it caused delays in the overall construction work. Due to this, as discussed above the project completion time was extended for several times".

Lack of construction material: The interviewees highlight lack of construction material as one important challenge. As per interviewees stated:

"There were lacks of construction materials in domestic market. Most of the materials needed for construction were not found in local market. Because of this most the construction materials were imported from China. In order to keep the specification and quality of the project, the required construction materials and inputs were imported from abroad mainly from the China".

More so, the interview also highlighted that related to these were the challenges faced by the customs of the importing countries. This had slowed down the importation of the required material. As a result, some delays had occurred.

Lack of skilled local manpower: the interviewees also mentioned the lack of skilled local manpower as another challenge. As per responses of interviewees there was shortage of the lack of qualified professionals in the country to perform certain tasks that required special skills and knowledge. One of the respondents shared the following regarding the Lack of skilled local manpower:

"There was critical shortage of skilled and experienced manpower locally in this project. Some of the standards and practices set out in the project documents were beyond the scope of construction in our country, so we needed manpower to understand those standards. There was shortage of the lack of qualified professionals in the country to perform certain tasks that required special skills and knowledge. As a result, these experts were brought in from abroad, and this took some time which in turn pose certain challenges in project". In this regards interviewee were mentioned that foreign expertise was specially involved in curtain work, wind design, landscape design, sound system, acoustics or other related tasks.

Lack of rules and regulations: The respondents were also mentioned that due to the complexity and scope of the project, challenges related to the rules and regulations were faced during the start of the project. One of the respondents shared the following regarding challenges related to the rules and regulations:

At the beginning of the project there were challenges related to lack of rules and regulations concerning the permit for construction of high-rise building. Since the building is very huge and one of the tallest in country, being built by an internationally experienced company and the project contract agreement has incorporated various international standards, at the beginning of the project there were various challenges associated with getting the construction permit and related licensing issues".

Challenges related to standards and practices: Another challenge associated with the project was that the difficulty to manage such huge mega project (high-rise building project) with our country construction building project management practices. There were many new elements, standards and practices that were beyond the scope of our country project management practices and standard. Interpreting, administering, and supervising such standards and such practices were also mentioned as challenges.

Language difference and barriers: The interviewees further indicated that language difference is another major challenge. Related to language difference and barriers, one of the interviewees mentioned the following:

"Language difference created challenges in understanding and interpreting some of the standards which are written in Chinese. There were so many standards in the project document that it took a long time to read, understand and interpretation of those standards. The project agreement was so vast, somehow unique and new to our country, so it was challenging to find a professional who could understand and interpret it quickly". **Challenges related to the management of claim and dispute**: The interview also mentioned that in terms of administering the project contact, there were some little challenges related with management of various claims. One of the respondents shared the following regarding challenges related to the management of claim and dispute:

"There were some claims by the contractors and managing and negotiating such claims were posing some little challenges as interviewees indicated. As interviewees were stated, practically, complaints are handled by negotiation between the Contractor and Employer/Consultant and nothing has gone to the Dispute Review Board or Arbitration so far. There were minor claims which were solved by discussion and negotiation before turning to disputes like; who owns facilities for Employer's Representative staff, Payment for Temporary Water, Electricity, detail finishing level and Partitions of inter. However, the Contract has Amicable Settlement, Dispute Adjudication Board and Arbitration as dispute settlement mechanism"

Overall, the study identified several challenges which among other include: the outbreak of COVID-19, lack construction materials, languages barrier, lack of some practices and standards, lack of skilled man power and contractor's claims

CHAPTER FIVE CONCLUSIONS AND RECOMMENDATIONS

Introduction

In this chapter, the conclusions and recommendations of the study were discussed. The project work was an objective to assess the practices of managing high rise building in Addis Ababa with the case of CBE head quarter building project. Based on this, the chapter presents the conclusion and recommendations that comprise further actions, which the study proposes for improving the project management practice of high-rise building project.

5.1. Summary of finding

As it is to be recalled, the major objectives of the research were to investigate the practices of managing high rise building in Addis Ababa with the case of CBE head quarter building project, to identify challenges of managing this project and to identified lessons learn from this project. To achieve these objectives, the study used literature review, semi-structure interview, and site observation as a research instrument where qualitative analysis was used to make discussion of the finding. In this section, the major findings of the research which have been discussed before are summarized in accordance with the objectives of the research. The study came to claim the following as key findings of the project work.

It was found from the study that from the various major project delivery methods in public and private construction projects, the contract type used for the CBE HQ project is the Design-Build (DB) delivery method. The main reasons for selecting this delivery method were that it is innovative method where the contractor takes full responsibility for the design and construction and it is suitable for large and complex projects like the CBE HQ building project; where early completion is a requirement.

In terms of time project management, the project had reported the time delay. In BD building contract, the Contractor takes the design risks and as such Request for Information (RFI's) and associated changes are minimal. However, the time management will consider the effect of excusable events on the schedule of the project. Due to the incorporation of additional element (scope) in project and unforeseen conditions associated with epidemics of Covid-19, the project

had been extended for four times for a total of 813 days. Over all, the project has taken 6 years and 5 months to complete.

Regarding the cost management practice, the study found that the contractor's offer under the DB route is a lump sum offer and this offer is little affected by price fluctuation. Due to the nature of DB building contract agreement (lump sum) the main responsibility of cost management mainly laid on the Contractor. Although scope and design changes after the approval are difficult to administer and changes will be expensive to implement in DB projects, based on the negotiation made between the employers and the contractors some changes were made on the project which leads to the increase in the project cost. Because of this, the project was completed with total cost of USD 303,476,558.17 from its initial contract value of USD 231,739,130.43 million.

The CBE used a Design-Build (DB) delivery method to execute its head quarter project. The project quality management process has set three quality milestones which are: CBE requirement, standards, and practices. The first of the essential features of a design and build contract is that the employer approaches a contractor with a set of requirements defining what the employer wants. The contractor responds with proposals, which include production as well as design work. The scale of design work included depends on the extent to which the employer has already commissioned such work from others. The various standards were set in contract and all the works were undertake according to standard set-in project agreement. In line with the stipulations of the design-build contract between the contractor and CBE, the consultant (AAiT) was responsible for managing any variation or change order requests made by respective contractual parties. AAiT review such requests and analyze their effect on the overall design and direct and indirect effect on the construction activities. The project quality management provided various mechanisms for AAiT to effectively address quality issues such as assessment and recommendation report, in depth supervision on the material, deficiency lists, design review, material testing at site laboratory & third-party Laboratory. AAiT prepare an assessment and recommendation report for CBE's and the contractor's team. Upon approval of such requests, AAiT follow up on the proper and effective implementation of such changes.

The study also identified major challenges concerning with management and implementation of this project. The study identified several challenges which among other include: the outbreak of COVID-19, lack construction materials, languages barrier, lack of some practices and standards, lack of skilled man power and contractor's claims. The study also identified that the project could have provided a significant opportunity for learning capacity and transferring technology and knowhow from the CSCEC Ltd; which is one of the most experienced contractors in the world.

In terms of project lessons learnt, the study found that throughout its construction, the project offered valuable lessons and witnessed important technological and experience-sharing endeavors. The CBE HQ building project has provided a significant opportunity for learning capacity for various interested party. The project had attracted significant visitors from academic institutes, government offices, and the industry, due to its complexity, use technological advanced construction methods, and overall significance to the Ethiopian construction. Various interested parties, professionals, and some college graduate and undergraduate students visited the project site as apprenticeships, reporting back to their respective universities on what they saw. In this regard, the project had provided a significant opportunity for learning capacity and transferring technology and knowhow. Because of the complexity and scope of the project, there were various technologies, standards and practices related with building of the high-rise building project which gives an important lesson and learning opportunities for local professional and Engineers. In this regard, curtain work, wind design, air conditioning, the elevator system, landscape design, sound system, lighting system, acoustics, and other associated activities were cited by interviewees as essential lessons for our country's construction project. Overall, Ethiopian engineering students and construction companies have benefited from the project by learning about cutting-edge high-rise building technologies.

5.2. Conclusion

The study has come a long way in discussing challenging and project management practices of the CBE head quarter project. Based on the finding of the study discussed above, the study makes conclusion. The study concludes that having skillful experts, documenting every project instruction in written form, providing discrete authority to the contractor with full responsibility (empowerment), harmonization of stakeholders, back-to-back evaluation with continuous progress review meetings were identified as how effective project term managed can be used to bring the project finally to be completed based on the experience of the case study. However, the study also conclude that from what have been said by the interview respondents, the effectiveness of the DB contract delivery system of the CBE HQ project on time, cost, quality and scope of work is summarized in the way that; this DB project was becoming effective in meeting its project cost though there were some change in cost were made due change in scope, however, the project was not effective in its time schedule as it has having significant time delay and the quality assurance is not certainly known though significant follow-up and supervision were made to ensure that project was in accordance with the details stated in the contract.

5.3. Recommendation

Based on the findings obtained from study, the study makes the following recommendations:

- Planning and scheduling of high-rise building construction is very important and complex. So, it is better the schedule prepared in detail and revised periodically by experienced professionals.
- Having a well-established contract is as good as having a good fence which leads to a smooth relationship with the stakeholders. It is hence recommended to have a clear contract first which later on leads to success with a proper follow up & monitoring of a project.
- Given the complexity and peculiar nature of the CBE HQ project, the consultant (AAU-AAiT) has to act as a key national institution for knowledge and technology transfer. All new technologies implemented in the project have to be disseminated to the local construction industry through various outreach mechanisms; including publication and seminars. A series of magazines on construction technology, new design approaches, instrumentation, electro mechanical systems and project management approaches have to be published and disseminated to local contractors, other consultants and even for college students of this field of study
- The number of skilled local workers in the electromechanical work and female workers in the project is less. Especially the number of skilled/professional local workers in the
electromechanical work better increased for the purpose of future maintenance, knowledge transfer and for the contractor it reduces salary cost in comparison with the foreigner.

• Trading off the three pillars of project management needs time to come up to a point of excellence. This means having well exposed and experienced contractors is beneficial for the most part. Though instead of bringing contractors from abroad, the local contractors can be brought to project management knowledge so that permanent solutions can be brought without incurring foreign currency outflows.

References

Agundu, P. Okwandu, G.and Owuala Mba, E. (2003). "Construction Project Management in Nigeria: Challenges and the Way Forward," Arc Construction Project Management.

Akinyele, A. (2016). "Determinants of Financial Technological Innovation Adoption by Customers of Deposit Money Banks in Nigeria". Jomo Kenyatta University of Agriculture and Technology, Kenya.

Ala-Risku, T. and Kärkkäinen, M. (2005). "Material Delivery Problems in Construction Projects: A Possible Solution," International Journal of Production Economics, in press.

Amalraj, J., Hernani, C., Ladouceur, K. and Verma, A. (2007). "Project Management: Challenges & Lessons Learned," BUEC 663, 2007.

Arditi, D. and Balci. G. (2009). "Managerial Competencies of Construction Managers," In Proceedings of the Fifth International Conference on Construction in the 21st Century Collaboration and Integration in Engineering, Management and Technology, May 20-22, Istanbul, Turkey, Paper 80, pp 650-658.

Ayalew, T. (2016). Assessment on Performance and Challenges of Ethiopian Construction Industry. Journal of Architecture and Civil Engineering, 2(11), pp.01-11.

Barkham, R. (1997): *The financial structure and ethos of property companies: An empirical analysis*, in: Construction Management & Economics, 15, 5, pp. 441-456.

Barkham, R. (2002): *Market research for office real estate*, in: Guy, S. & Henneberry, J.: Development and developers: Perspectives on property, Blackwell Publishing, pp. 53-72.

Barkley, B.T. (2004). Project Risk Management. McGraw-Hill companies, Inc.

Beedle, L.S. (2006). Tall building structures – a world review. Council on tall buildings and urban habitat

Bowen, A., Hall, A., Edwards, J., Pearl, G. and Cattell, S. (2002). "Perceptions of Time, Cost and Quality Management on Building Projects," The Australian Journal of Construction Economics and Building, 2(2), pp. 48-56.

Byrne, P. J. (1996). Risk, uncertainty and decision-making in property development, 2nd ed, E. & FN Spon.

Chartered Institute of Building. (2002). Code of Practice for Project Management for Construction and Development. 3rd edition. Blackwell Publishing.

Declerck, R., Debourse, J. & Navarre, C. (1983). La Méthode de Direction générale: le management stratégique. Paris: Hommes et Techniques.

Easterby-Smith M, Thorpe R and Jackson P (2012). Management Research (4th ed.). London: SAGE Publications Ltd.

Emporis. (2000). www.emporis.com. Retrieved 4 26, 2021, from emporis: www.emporis.com/building/standard/3/hig-rise-building

Fapohunda, J. and Stephenson, P. (2010). "Optimal Construction Resources Utilization: Reflections of Site Managers' Attributes," The Pacific Journal of Science and Technology, 11(2), pp.353-365.

Frezewd, A. (2016). Practice of Project Risk Management in Batu and Dukem Town water supply Project. Msc Thesis. Addis Ababa, Ethiopia: Addis Ababa University.

GCR (2020). Global Competitiveness Report Special Edition 2020: How Countries are Performing on the Road to Recovery. World Economic Forum

Gehner, E (2008): Knowingly taking risk - Investment decision making in construction building development, eburon.

Gehner, E., Halman, J. & De Jonge, H. (2006). Risk management in the Dutch real estate development sector: a survey; in: Amaratunga, D. / Haigh, R. / Vrijhoef, R. / Ham: Amaratunga, D., Haigh, R., Vrijhoef, R., Hamblett, M. & Van den Broek, C. (Eds.): Proceedings of the 6th International postgraduate research conference in the built and human environment, Delft, The Netherlands, 6-7 April 2006 (pp. 541-552).

Getachew, E.T. (2018). Assessing Project Management Practices of Prepaid Energy Metering Project in Ethiopian Electric Utility: The Case of Addis Ababa. Master of Arts in Project Management.

Getnet, A. (2019). Assessment of Project Risk Management Practice: The case study of Commercial Bank of Ethiopia Information Technology Infrastructure Library (ITIL) Project: Project Work. Addis Ababa, Ethiopia: Addis Ababa University.

Giang D. T & Pheng S.L. (2010). Role of Construction in Economic Development: Review of key in the past 40 years.

Gorecki, J. (2015) Maturity of Project Management in Polish and Foreign Construction Companies. Foundations of Management, Vol. 7, pg. 71-82.

Grant, K.P. and Pennypacker, J.S. (2006). Project Management Maturity: An Assessment of Project Management Capabilities Among and Between Selected Industries. IEEE Transactions on Engineering Management, Vol. 53, No. 1, pg. 59-68.

Hai, T., Yusof, M., Ismail S. and Wei, F. (2012). "A Conceptual Study of Key Barriers in Construction Project Coordination," Journal of Organizational Management Studies. Vol. 2012.

Hailu, G. (2019). Project Management Practices of Ethiopian Electric Power Corporation: The Case of Universal Electricity Access Program (UEAP). Master of Arts in Project Management.

Hendrickson, C. (2008). Project Management for Construction: Fundamental Concepts for Owners, Engineers, Architects and Builders. 2ndedition. Pittsburgh, PA: Carnegie Mellon University.

Jacob, B. and McClelland, T. (2001). "Theory of Constraints Project Management-A Brief Introduction into the Basics," The Goldratt Institute.

Kar, D. (2007). "An Innovative Methodology for Arresting Project Slippage in Water Resources and other Infrastructural Projects," Doctoral Thesis to Bengal Engineering and Science University, Shibpur, Howrah 711 103, INDIA. Kar, D. (2009). "Implementation Construction Projects on Schedule -Challenge in a Developing Economy," Journal of Economics and International Finance, 1(4), pp.088-092.

Kerzner, H. (2003). Project management, A systems approach to planning, scheduling and controlling. New York: John Wiley and Sons.

Kerzner, H. (2006). Project management best practices: Achieving global excellence. Hoboken, NJ: John Wiley & Sons. Kerzner, H. (2009). Project Management (10th ed.).

Kerzner, H. (2017). Project Management (12th ed.).

Kerzner, R. (2011). Using the project management maturity model: strategic planning for project management. New York: John Wiley & Sons.

Kombo, D. & Tromp, D. (2009). Proposal and Thesis Writing; An Introduction. Nairobi, Kenya: Don Bosco Printing Press.

Koskela L & Howell (2002). "The underlying theory of project management is obsolete"

Kothari, C. (2004). Research Methodology. Methods and Techniques. (2nded). India, New Delhi: New age International Publishers.

Lavrakas, P. (2008). Encyclopedia of Survey Research Methods. United States of America, Los Angeles; Sage Publications.

Lester, A. (2017). Project Management, Planning and Control Managing Engineering, Construction and Manufacturing Projects to PMI, APM and BSI Standards. 7th ed. United Kingdom and United States: Elsevier.

Lewis, James P. (2011). Project Planning, Scheduling and Control. (5th edn). McGraw-Hill companies, Inc.

Manalebih, M. (2018). Assessing the Practice of Project Risk Management: The case of World Vision Ethiopia Water, Sanitation and Hygiene (WASH) projects, Project Work. Addis Ababa, Ethiopia: Addis Ababa University.

Miles, M. E., Haney, R. L. & Berens, G. (1995): Real Estate Development: Principles and Process, Urban Land Institute.

Millington, A. F. (2000): Property development, Estates Gazette.

Muir, B. (2005). "Challenges Facing Today's Construction Manager," Supplemental Reading for CIEG 486-010, Construction Methods & Management, University of Delaware.

Mutchler, R. and Widener, C. (2000). "Construction Management, Excerpt from The Architect's Handbook of Professional Practice," 13th edition. Supplemental Architectural Services, AIA.

National Bank of Ethiopia (NBE) (2019). The 2018/19Annual Report. Retrieved from www.nbebank.com.

Neary, N. (2009). Growth strategies for real estate developers - Motives, experiences and strategies in entering foreign markets, MSRE Master Thesis, Amsterdam School of Real Estate, May 2009.

Negera, B. (2018). Assessment on the Project management practice of Ethiopian Petroleum Supply Enterprise (EPSE) Application and Network Infrastructure Project. Degree of Master of Arts in Project Management.

Newing, H. (2011). Conducting research in Conservation: Social Science Methods and Practice. USA, New York: NY Route ledge.

Nozeman, E.F. & Dorenbos, R.J. (2006): The European real estate development industry: Structure and dynamics, Paper prepared for the 13th Annual European Real Estate Society Conference, Weimar, Germany, 7-10 June 2006.

Othman, E. (2013). "Challenges of mega construction projects in developing countries," Organization, Technology and Management in Construction an International Journal, 5(1), pp.730-746.

Parker, L. (2003). Qualitative Research in Accounting and Management: The emerging Agenda. Journal of Accounting and Finance, Vol.2, 15-30.

Pinto (2007) Field guide to project management 2nd edition.

PMI (1994). A guide to the Project Management Body of Knowledge. Newtown Square, PA: Project Management Institute.

PMI (2000). A Guide to the Project Management Body of Knowledge. Newtown Square, PA: Project Management Institute.

Project Management Institute (2017). A guide to the project management body of knowledge. 6th ed. Pennsylvania: Project Management Institute.

Project Management Institute (PMI), (2004). A Guide to the Project Management Body of Knowledge (PMBOK Guide), Third Edition, Pennsylvania, United States.

Project Management Institute. (2003). Organizational Project Management Maturity Model (OPM3). Newtown Square, Pennsylvania: Project Management Institute, Inc.

Project Management Institute. (2013). A Guide to the Project Management Body of Knowledge. 5thedition. Newtown Square, Pennsylvania: Project Management Institute, Inc.

Project Management Institute. (2016). Construction Extension to the PMBOK Guide. Newtown Square, Pennsylvania: Project Management Institute, Inc.

Project Management Professionals Certification Center (2002). A Guidebook for Project and Program Management for Enterprise Innovation. Japan: Project Management Professionals Certification Center.

Robson, S. (2009). The use of judgment by commercial property developers, RICS COBRA Research Conference, University of Cape Town, 10-11th September 2009.

Saatcioglu, Murat (2016). "High-Rise Buildings in Natural Disaster". Encyclopedia of Natural Hazards. Encyclopedia of Earth Sciences Series. Dordrecht, NL: Springer. pp. 451–452

Saunders M, Lewis P and Thornhill A (2009) Research Methods for Business Students (5th ed.). Essex, England: Pearson Education Limited.

Schwab, D. (2005). Research Methods for Organizational Studies. (2nded.) London: Lawrence Erlbaum Associates.

Temesgen, T. W. (2013). Assessment of project management capability: Case study at Mesfin Industrial Engineering PLC, Mekelle –Ethiopia). Master thesis, Royal Institute of Technology (KTH).

Tuman, G.J. (1983). Development and implementation of effective project management information and control systems, in Cleland, D.I. & King, W.R. (eds.) Project management handbook. New York: Van Nostrand Reinhold Co., 495-532.

Turner, J.R. (1999) The handbook of project-based management: improving the processes for achieving strategic objectives. 2nd ed. London: McGraw-Hill.

Walker, A. (2002/2015). Project Management in Construction, 4th edition, Oxford, UK, Blackwell Science Ltd.

Wiegelmann, T. (2012). "Risk Management in the Real Estate Development Industry: Investigations into the application of risk management concepts in leading European real estate development organizations. Doctoral Thesis, Institute of Sustainable Development & Architecture Bond University, Australia.

Wilkinson, S. & Reed, R. (2008): Property Development, 5th ed., Routledge.

Wysocki, R.K. (2014). Effective Project Management Traditional, Agile and Extreme. 7th ed. Indiana and Canada: John Wiley & Sons.

Yimam, A.H. (2011). Project Management Maturity in the Construction Industry of Developing Countries. Master's thesis, University of Maryland.

Zelele, T. (2016). Assessment on Performance and Challenges of Ethiopian Construction Industry. Journal of Architecture and Civil Engineering, Vol. 2, pp., 1-11.

Appendix I: Interview questions

The purpose of this interview is to collect data for the study entitled "Assessment of Project Management Practices in High-Rise Building: The Case of CBE Headquarter Building Project " for partial fulfillment of the requirements for the Master of Art (MA) in Project Management in St. Merry University. This interview is required to assist in determining the objectives of the study. Your participation in this survey is voluntary. The information you provide will be used only for the purpose of the study and will be kept strictly confidential. Therefore, you are kindly requested to provide your genuine responses to different questions followed. Thanks in advance for your cooperation for this interview.

Section I: General Questions

- 1. What is your responsibility/position in the project?
- 2. What is your responsibility in the project?
- 3. How long you are working in this project?

Section: II:

- 1. Let us briefly explain the overall status of CBE head quarter project? How do you evaluate the status of CBE head quarter project against objectives?
- 2. Let me briefly explain the overall project management practices of CBE head quarter project? What do you think the purpose and objective of project management? Does the project clearly establish the purpose and objective of overall project management? If yes, is it formally communicated to all project stakeholders?
- 3. Are project management role and responsibilities assigned to project stakeholders? How?
- 4. How do you observe the project cost management practice in CBE head quarter project? Is there project cost overrun in this project? What were the major challenges in managing cost of the project? How did you overcome those challenges?
- 5. How do you observe the project time schedule management practice in CBE head quarter project? Is the timeline for performing project management activities set? How? Is there any delay in project time schedule? If yes, for how long it delays? What were the major challenges faced in managing time schedule of the project? How did you overcome those challenges?

- 6. How do you observe the project quality management practice in CBE head quarter project? Were there any scope changes with CBE head quarter project? If yes, how was the scope change handled?
- 7. Over all, what are the main challenges in managing CBE headquarter projects proficiently? What kind of solutions did you give for those problems and challenges? How do you handle them?
- 8. Is there any practice of lesson learned from of the project? If yes, what were the strengths and weakness mentioned in practicing project management?

Thank You Once Again for Your Cooperation and Helpfulness!